

EIOPA-BoS-21-234 10 June 2021

Decision of the Board of Supervisors

on the collaboration of the insurance supervisory authorities of the Member States of the European Economic Area

THE BOARD OF SUPERVISORS OF THE EUROPEAN INSURANCE AND OCCUPATIONAL PENSIONS AUTHORITY,

Having regard to Regulation (EU) No 1094/2010 of the European Parliament and of the Council of 24 November 2010 establishing a European Supervisory Authority (European Insurance and Occupational Pensions Authority), amending Decision No 716/2009/EC and repealing Commission Decision 2009/79/EC¹, in particular Article 29 thereof,

Whereas:

Articles 152a and 152b of Directive 2009/138/EC of the European Parliament and of the Council of 25 November 2009 on the taking-up and pursuit of the business of Insurance and Reinsurance (Solvency II)² (as amended) introduce notification requirements and collaboration platforms in the regulatory framework with effect of 30 June 2021. Therefore, the Decision of the Board of Supervisors on the collaboration of the insurance supervisory authorities of the Member States of the European Economic Area (EIOPA-BoS-17/013) of 30 January 2017 should be revised accordingly. In addition, the revision needs to take into account the outcome and recommended actions of EIOPA's Peer Review on EIOPA's Decision on the collaboration of the insurance supervisory authorities and the experience gained in the application of that Decision.

HAS DECIDED AS FOLLOWS:

¹ OJ L 331, 15.12.2010, p 48.

² OJ L 335, 17.12.2009, p. 1.

Article 1

- 1. Supervisory authorities shall perform the notifications and information exchange set out in the Annex to this Decision by e-mails until EIOPA develops an appropriate IT tool for this purpose.
- 2. EIOPA shall provide an email address to the supervisory authorities to enable them to perform the notifications and information exchange.

Article 2

If certain supervisory tasks, as laid down in Directive 2009/138/EC, fall under the competence of other supervisory authorities that are not members of the Board of Supervisors, those supervisory authorities may, subject to approval by the Board of Supervisors, apply this Decision by signing a declaration of commitment, if this contributes to further strengthening cross-border cooperation.

Article 3

The Decision of the Board of Supervisors on the collaboration of the insurance supervisory authorities of the Member States of the European Economic Area (EIOPA-BoS-17/013) shall be repealed with effect from 1 July 2021.

Article 4

This Decision shall enter into force on the day following its adoption and shall apply from 1 July 2021.

Article 5

- 1. With regard to the events started prior than 1 July 2021 which would have triggered a notification according to this Decision, supervisory authorities shall evaluate, on a case by case basis, to perform a notification taking into account the specific circumstances in the context of cross-border supervision and the expected timeframe for the situation to be resolved.
- 2. Supervisory authorities shall perform notifications based on that assessment by 30 September 2021

Done at Frankfurt am Main, 10 June 2021

For the Board of Supervisors
Peter Braumüller
Vice-Chairperson

ANNEX TO BOS DECISION ON THE COLLABORATION OF THE INSURANCE SUPERVISORY AUTHORITIES OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC AREA (EIOPA-BOS-21-234)

EIOPA-BoS-21-235 10 June 2021



Table of Contents

1	PAF	RT I GENERAL CONSIDERATIONS	.5
	1.1	General aims	.5
	1.2	Principles of cooperation	.6
	1.3	Rules on professional secrecy	.6
	1.4	IT tool supporting the exchange of information and notifications	. 7
2	PAF	RT II AUTHORISATION	.8
	2.1	General	.8
	2.2	Exchange of authorisation information with other insurance Supervisory	
	Autho	orities	.8
	2.2.1	Scope	.8
	2.2.2	Addressee	.9
	2.2.3	Information	
	2.2.4	Providing of information	10
	2.3 which	Exchange of authorisation information with financial supervisory authorities are not insurance supervisory authorities	10
	2.3.1	Scope	10
	2.3.2	Addressee	10
	2.3.3	Information	10
	2.4 hold	Exchange of information on all persons who effectively run the undertaking on the three controls, shareholders and members with qualifying holdings	
	2.5	Exchange of information on authorisations sought in other Member States	13
	2.6	Notification on cross-border activity	13
	2.6.1	Scope	13
	2.6.2	Addressee	14
	2.6.3	Information	14
	2.6.4	Assessment of the notification	15
	2.7	Acquisitions and increase of shareholdings in undertakings	16
3	PAF	RT III CROSS-BORDER ACTIVITIES	18
	3. Fra	mework for the collaboration of Supervisory Authorities	18
	3.1 (re)ir	Establishment, change of information and closure of a branch by a(n) surance undertaking	19
	3.1.1 EIOP	Information to be communicated by the Home NSA to the Host NSA and A in case of insurance branch establishment	19
	3.1.2 case	Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to be provided by the Host NSA to the Home NSA and EIOPA in the Information to the Information t	

	3.1.3 in case	of insurance branch establishment	_
	3.1.4	Starting the activity of the branch	. 24
	3.1.5	Changes to information concerning the branch	. 24
	3.1.6	Branch closure	. 25
	3.1.7	Establishment, change of information and closure of a branch by a	
		ance undertaking	
		commencing activities by way of freedom to provide services by an insurantaking	
	3.2.1 EIOPA	Information to be communicated by the Home NSA to the Host NSA and 27	
	3.2.2	Information to be provided by the Host NSA to the Home NSA and EIOPA.	. 30
	3.2.3 underta	Information to be provided by the Home and Host NSAs to the insurance aking	. 31
	3.2.4	Starting of activity by the insurance undertaking	. 32
	3.2.5	Changes to information	. 32
	3.2.6	Cessation of Activities	. 32
	3.3 C	Other provisions common to branches and freedom to provide services	. 32
	3.3.1	Language, communication means, contact points and lists	. 32
	3.3.2 alterna	Submission of policy conditions to the Host NSA (health insurance as an tive to social security)	. 34
	3.3.3	Representative for the handling of claims (motor vehicle liability)	. 34
4	PART	IV SUPERVISION ON A CONTINUOUS BASIS	. 36
	4.1 C	continuing cooperation between Home and Host NSAs (branches and FoS)	. 36
	4.2 E	xchange of information	. 37
	4.2.1	Notifications from the Home NSA	. 37
	4.2.2	Other exchange of information from the Home NSA	. 43
	4.2.3	Notification from the Host NSA	. 45
	4.2.4	Other exchange of information from the Host NSA	. 47
	4.3 C	college of Supervisors and collaboration platforms	. 50
	4.3.1	Where a college of supervisors is established	. 50
	4.3.2	Collaboration platforms	. 51
	4.4 P	ortfolio transfer	. 53
	4.4.1	Transfer of portfolio of contracts of insurance undertakings	. 53
	4.4.2	Transfer of portfolio of contracts of reinsurance undertakings	. 56
	4.4.3	Transfer of portfolio of contracts of insurance branches of Third Countries	58

	4.5	Supervisory measures	. 59
	4.5.1	Restriction or prohibition of free disposal of assets	. 59
	4.5.2	Recovery plan	. 59
	4.5.3	Short-term finance scheme	. 60
	4.5.4	Withdrawal or lapse of authorisation	. 60
	4.5.5 insura	Application of reorganisation measures and winding-up proceedings of an ance undertaking	
5	PAF	RT V REGULAR EXCHANGE OF QUANTITATIVE DATA	. 62
	5.	Exchange of quantitative information between the Home NSA and Host NSA	62
6	PAF	RT VI HANDLING OF POLICYHOLDER COMPLAINTS	. 64
	6.1	Complaints handling systems in Member States	. 64
	6.2	Complaints handling procedures	. 65
	6.3	Dispute over responsibilities	. 66
	6.4	Information to the public	. 66
7	PAF	RT VII OTHER ISSUES	. 68
	7.1	Assistance (class 18)	. 68
	ANNE	XES	. 69
	ANNE	X 1: Certificate of Solvency	. 69
		EX 2: Table summarising the information flows between the Supervisory prities foreseen by this Decision	. 71
	ANNE	X 3: Information included in the templates used for notifications	82

1 PART I GENERAL CONSIDERATIONS

1.1 General aims

- 1.1.1 This Decision applies to the authorities of the Member States of the European Economic Area (EEA) competent for the supervision of insurance and reinsurance undertakings ("Supervisory Authorities") on the basis of Directive 2009/138/EC¹ ("Solvency II Directive").
- 1.1.2 This Decision should be read together with the relevant provisions of the Solvency II Directive.
- 1.1.3 The Members of the European Insurance and Occupational Pensions Authority ("EIOPA") have a long tradition of cooperation and exchange of information. It is recognized that the on-going development of the internal market and the growing internationalisation of business activities require increased collaboration between supervisors.
- 1.1.4 The Supervisory Authorities will endeavour to cooperate and use, as effectively as possible, all information available for supervisory purposes in order to achieve the objectives of insurance supervision and, in particular, adequate protection of policyholders and other stakeholders and financial stability.
- 1.1.5 Nothing in this Decision should affect internal communication arrangements within a Member State. However, such arrangements should not hinder effective cross-border cooperation between the Supervisory Authorities.
- 1.1.6 Nothing in this Decision should diminish the responsibilities of Supervisory Authorities under the Solvency II Directive and in particular the duty vested in the Supervisory Authority(ies) of the Home Member State to authorise and to exercise prudential supervision over the undertaking for which it has sole responsibility.

¹ Directive 2009/138/EC of the European Parliament and of the Council of 25 November 2009 on the pursuit of the business of insurance and reinsurance (Solvency II) (OJ L 335, 17.12.2009, p.1)

1.2 Principles of cooperation

- 1.2.1 The Supervisory Authorities, whilst recognising the principles in the Solvency II Directive regarding the distribution of responsibilities between Home and Host Member State Supervisory Authorities ("Home NSA" and "Host NSA" respectively) acknowledge the importance of effective cooperation to facilitate the carrying out of their respective duties.
- 1.2.2 The procedures and information described in the Decision constitute minimum requirements and provide considerations on the communication of additional information. Nothing in the Decision should be interpreted in such a way as to hinder the exchange of further information and the extended collaboration between the Supervisory Authorities, as concerns the supervision of undertakings.
- 1.2.3 The information exchange requirements described in the Decision should be proportionate to the nature, scale and complexity of the risks inherent in the business of the insurance or reinsurance undertaking. The Supervisory Authorities should give adequate consideration to the specific nature of captive insurance and captive reinsurance undertakings.
- 1.2.4 The information exchange requirements described in the Decision should take into account prospective and risk-based supervision which includes the verification on a continuous basis of the proper operation of the insurance or reinsurance business and of the compliance with supervisory provisions by insurance and reinsurance undertakings.
- 1.2.5 In the case of undertakings in difficulty and other emergency situations, the Supervisory Authorities endeavour to make every effort to remedy the problems and minimise the negative impact on policyholders and financial stability.
- 1.2.6 In the case of divergent views in terms of application of this Decision, the relevant Supervisory Authorities should strive to a mutually agreed solution; in cases where no such solution can be reached, recourse should be made to EIOPA for mediation.

1.3 Rules on professional secrecy

The Supervisory Authorities agree to exchange confidential information whenever possible, within the limits of the rules laid down in Articles 64 to 71 of the Solvency II Directive in order to improve the effectiveness of insurance and reinsurance supervision in the EEA.

1.4 IT tool supporting the exchange of information and notifications

- 1.4.1 To allow the identification of potential risks as soon as possible, the exchange of information and notifications prescribed in this Decision in relation to undertakings pursuing cross-border activity under the freedom to provide services (FoS) and freedom of establishment (FoE) will be supported by a dedicated IT tool.
- 1.4.2 The IT tool is provided and hosted by EIOPA and it is made available to each Supervisory Authority. The IT tool is used:
 - as platform to share information between the Home and the Host NSAs and EIOPA;
 - as data storage system to enable structured storage and extraction of the information shared on an individual and aggregated basis (database).

EIOPA operates the IT tool in accordance with applicable data transmission and data protection rules.

- 1.4.3 For each information exchange and notification in relation to undertakings pursuing cross-border business specific templates will be designed. Each template contains the minimum set of information to be exchanged by the Supervisory Authorities. Annex 2 to this Decision describes the information flows between the Supervisory Authorities foreseen by this Decision with the names of the related template. Annex 3 to this Decision contains the information included in the templates.
- 1.4.4 With the aim to continuously improve the processes of information and notification exchange between Supervisory Authorities including the time and the quality of information exchanged, EIOPA may conduct reviews of the process and of the information exchanged through the IT tool. The Supervisory Authorities should report to EIOPA any improvement opportunities they detect to improve both the process and the information exchanged.

2 PART II AUTHORISATION

2.1 General

Article 15 of the Solvency II Directive

- 2.1.1 In accordance with the principle of single authorisation, the decision to issue an authorisation, which is valid for the whole of the EEA, shall be the sole responsibility of the Home NSA.
- 2.1.2 Within the context of the single market and with due regard to the single licence principle, the Supervisory Authorities recognise the importance of conducting a rigorous examination of the authorisation information, and of seeking to ensure the uniform application of the criteria relating to the supervision of access to, and pursuit of, insurance and reinsurance activity.
- 2.1.3 The Home NSA shall upload on its website a list of all insurance and reinsurance undertakings licensed in that Member State as well as third country branches, including at least, the name, postal address, legal form and authorised insurance classes. The information shall be updated as necessary².

2.2 Exchange of authorisation information with other insurance Supervisory Authorities

Article 26(1) and (3) of the Solvency II Directive

2.2.1 Scope

2.2.1.1 Section 2.2 is applicable when the undertaking requesting the authorisation is:

- a) a subsidiary of an insurance or reinsurance undertaking authorised in another Member State;
- b) a subsidiary of the parent undertaking of an insurance or reinsurance undertaking authorised in another Member State;
- c) an undertaking controlled by the same person, whether natural or legal, who controls an insurance or reinsurance undertaking authorised in another Member State.

² See 3.3.1.5 for disclosures requirements applicable from a Host perspective.

2.2.2 Addressee

- 2.2.2.1 The Supervisory Authority responsible for granting authorisation prior to the authorisation shall ask for the information defined in paragraph 2.2.3 from:
 - a) the Supervisory Authority of the direct parent undertaking referred in 2.2.1.1 a) and, if applicable, of the ultimate parent undertaking at EEA level,
 - b) the Supervisory Authority of the subsidiary and the group supervisor under the Solvency II Directive;
 - c) the Supervisory Authority of the undertaking controlled by the same person, whether natural or legal, in the other Member State.

2.2.3 Information

- 2.2.3.1 The Supervisory Authority responsible for granting authorisation shall ask for any relevant information relating to:
 - a) the suitability and the financial soundness of the shareholders and members with qualifying holdings;
 - b) the structure of the group;
 - c) the fitness and propriety assessments of all persons who effectively run the undertaking or hold other key functions, if these persons are involved or have been involved in the management of another entity of the same group;
 - d) deteriorating financial conditions and instances of non-compliance with technical provisions, Solvency Capital Requirement ("SCR") and Minimum Capital Requirement ("MCR") of the relevant insurance or reinsurance undertakings and measures taken in accordance with Articles 137, 138, 139 and 141 of the Solvency II Directive;
 - e) significant concerns in terms of the system of governance including risk management of the relevant insurance or reinsurance undertakings;
 - f) supervisory measures taken under Articles 36 and 37 of the Solvency II Directive to address deficiencies related to the system of governance of the relevant insurance or reinsurance undertakings;
 - g) possible conflicts of interest;
 - h) other relevant licensing criteria, as appropriate.
- 2.2.3.2 If the ultimate parent of the undertaking seeking authorisation is not a regulated undertaking, or is a regulated undertaking located in a third country, the Supervisory Authority of the ultimate parent undertaking at EEA level or, if applicable, the group supervisor of the EEA group, shall be asked to provide the requesting Supervisory Authority with the information listed in paragraph 2.2.3.1.

2.2.4 Providing of information

2.2.4.1 The Supervisory Authority to whom the request has been made, shall as soon as possible and preferably within 2 weeks upon receipt of a request, provide the Supervisory Authority responsible for granting the authorisation with any relevant information in its possession or inform the requesting authority that it does not have any such information.

2.3 Exchange of authorisation information with financial supervisory authorities which are not insurance supervisory authorities

Article 26(2) and (3) of the Solvency II Directive

2.3.1 Scope

- 2.3.1.1 Section 2.3 is applicable when the undertaking requesting the authorisation is:
 - a) a subsidiary of a credit institution or investment firm authorised in another Member State,
 - b) a subsidiary of the parent undertaking of a credit institution or investment firm authorised in another Member State,
 - an undertaking controlled by the same person, whether natural or legal, who
 controls a credit institution or investment firm authorised in another Member
 State.

2.3.2 Addressee

- 2.3.1.2 The Supervisory Authority responsible for granting authorisation prior to the authorisation shall ask the information defined in paragraph 2.3.3 respectively from:
 - a) the Supervisory Authority of the direct parent undertaking and, if applicable, of the group supervisor of the EEA group,
 - b) the Supervisory Authority of the subsidiary,
 - c) the Supervisory Authority of the undertaking controlled by the same person, whether natural or legal, in the other Member State.

2.3.3 Information

- 2.3.1.3 The Supervisory Authority responsible for granting authorisation shall ask for any relevant information relating to:
 - a) the suitability and the financial soundness of the shareholders and members with qualifying holdings;
 - b) the structure of the group;

PART II AUTHORISATION

- c) the fitness and propriety assessments of all persons who effectively run the undertaking or hold other key functions, if these persons are involved or have been involved in the management of another entity of the same group;
- d) deteriorating financial conditions and instances of non-compliance with quantitative sectoral requirements by the relevant credit institutions or investment firms:
- e) significant concerns in terms of the system of governance of the relevant credit institutions of investment firms including risk management;
- f) supervisory measures taken in the case described in point d) and to address deficiencies related to the system of governance of the relevant credit institutions or investment firms;
- g) possible conflicts of interest;
- h) other relevant licensing criteria, as appropriate.
- 2.3.1.4 If the ultimate parent is not a regulated undertaking, or is a regulated undertaking located in a third country, the Supervisory Authority of the ultimate parent undertaking at EEA level or, if applicable, the group supervisor of the EEA group, shall be asked to provide the requesting Supervisory Authority with the information listed in paragraph 2.3.3.1.

2.4 Exchange of information on all persons who effectively run the undertaking or hold other key functions, shareholders and members with qualifying holdings

Article 26 (3) of the Solvency II Directive

- 2.4.1 With regard to the importance of the supervision of
 - a) the fitness and propriety assessment of all persons who effectively run the undertaking or hold other key functions, if these persons are involved or have been involved in the management of another entity in the same group; and
 - b) the suitability and the financial soundness of the shareholders and members with qualifying holdings;
 - both during the process of authorisation and on an on-going basis, the Supervisory Authorities shall endeavour to exchange available information related to these areas.
- 2.4.2 Where the persons who effectively run the undertaking or hold other key functions and/or the shareholders and members with qualifying holdings come

PART II AUTHORISATION

from, or are connected to³ other Member States, the Supervisory Authority concerned shall request, where applicable, from the Supervisory Authorities of those other Member States, relevant information relating to:

- a) the fitness and propriety assessments of all person(s) who effectively run the undertaking or hold other key functions; and
- b) the suitability and financial soundness of the shareholders and members with qualifying holdings.
- 2.4.3 The Supervisory Authority to which the request has been made, shall, as soon as possible, preferably within 2 weeks upon receipt of a request, provide to the Supervisory Authority responsible for granting authorisation, any relevant information in its possession or inform the requesting Supervisory Authority that it does not have any such information.
- 2.4.4 The Home NSA shall immediately inform the Host NSA when it takes any measure against person(s) who effectively run the undertaking or hold other key functions, or against its shareholders and members with qualifying holdings⁴.
- 2.4.5 In accordance with Article 26(3) of the Solvency II Directive, the Home NSA shall inform the Host NSA concerned of all decisions that might be of relevance⁵ to the Host NSA when assessing
 - a) the fitness and propriety of all person(s) who effectively run the undertaking or hold other key functions if these persons are involved or have been involved in the management of another entity of the same group;
 - b) the suitability and the financial soundness of the shareholders and members with qualifying holdings.
- 2.4.6 The Supervisory Authorities shall establish a contact point or set up a general email account for the exchange of information on the suitability of shareholders and members with qualifying holdings and on the fit and proper requirements of all persons who effectively run the undertaking or hold other key functions.
- 2.4.7 EIOPA shall develop an appropriate tool that makes the above contact points available to the Supervisory Authorities. This shall be hosted on the EIOPA

³ For the purpose of assessing " come from, or are connected to other Member States", the concerned Supervisory Authority shall consider the communicated experience of that person(s), namely if the CV indicates that the person has worked in a regulated entity in other Member States.

⁴ The notion of "any measure" shall cover not only administrative measures but also misdemeanours and criminal proceedings if applicable such as when the Supervisory Authority is the e.g. authorized prosecutor.

⁵ It shall cover relevant decisions issued or known by the Supervisory Authority taking into account the provisions of professional secrecy.

extranet. The Supervisory Authorities shall be responsible for keeping that information up to date.

2.5 Exchange of information on authorisations sought in other Member States

- 2.5.1 The Home NSA shall request the applicant to declare if there had been a formal or informal request for an authorisation, by its shareholders or members with qualifying holdings, to establish an insurance or reinsurance undertaking in another Member State or third country that had been rejected or withdrawn. The Home NSA shall ask the applicant for the reasons why the application was rejected or withdrawn.
- 2.5.2 Where appropriate, the Home NSA shall engage with the Supervisory Authority(ies) from whom the application has been sought in order to understand the circumstances of the rejected or withdrawn application, before making a decision on the authorisation.
- 2.5.3 The Supervisory Authorities shall provide input in a timely manner, preferably within 2 weeks upon receipt of a request, with a view to exchanging information before the decision on the particular application is taken by the Home NSA.
- 2.5.4 The Supervisory Authorities shall keep record (preferably in a database) of the rejected and withdrawn authorisation requests.

2.6 Notification on cross-border activity

Article 152a(1) of the Solvency II Directive

2.6.1 Scope

- 2.6.1.1 Section 2.6 is applicable when the scheme of operations of the undertaking requesting the authorisation indicates that:
 - a) a part of its activities will be based on the freedom to provide services, in case of insurance undertakings, or the freedom of establishment, in case of insurance and reinsurance undertakings, in another Member State; and
 - b) those activities are likely to be of relevance with respect to the host Member State's market.
- 2.6.1.2 For the purpose of point b) of the previous paragraph, the activities of an undertaking are considered relevant with respect to the host Member State's market when the gross written premiums in the host Member State indicated in the scheme of operations is higher than EUR 0.

2.6.2 Addressee

- 2.6.2.1 The Supervisory Authority responsible for granting authorisation prior to the authorisation shall notify the information defined in paragraph 2.6.3 to:
 - a) EIOPA,
 - b) the Supervisory Authorities in charge of Prudential and Conduct of business supervision in the host Member State.

2.6.3 Information

- 2.6.3.1 When the undertaking states it intention to start cross-border activity in the future⁶, the notification shall contain the following information::
 - a) the name and address of the undertaking;
 - b) where applicable, the LEI of the undertaking (or if not available, the identification code used by the Supervisory Authority);
 - c) where applicable, information of the and/or the branches or other establishments in the host Member States from which it plans to provide services:
 - d) the classes of insurance according to Annexes I and II to the Solvency II Directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive which the insurance undertaking is applying for authorisation;
 - e) a summary of the undertaking scheme of operations indicating:
 - a. the nature of the risks or commitments which the undertaking proposes to cover in the Host Member State(s);
 - b. if available, the projected gross written premiums, estimates of management expenses, general expenses and commissions; guiding principles of the reinsurance retrocession of the business, projected loss ratio of the portfolio and other projected technical indicators if available, for each Host Member State;

⁶ The undertaking plans to start cross-border activity within the first three financial years in accordance with the scheme of operations.

- c. if available, any information regarding local service provider involved in the underwriting activities in the Host Member State⁷ including but not limited to potential distribution partners in the host Member State:
- f) where the undertaking has clearly indicated its intention to operate exclusively, or almost exclusively, in the Host Member State(s):
 - a. identification, through for example the person(s) ID card, passport or nomination document(s), of the persons who effectively run the undertaking or are responsible for the key functions;
 - b. a summary of the undertaking's system of governance, including the risk management system in place;
 - c. any available information resulting from the discussions with the undertaking of its business strategy and how the activities in the host Member State fits into that strategy.
- 2.6.3.2 When the undertaking notifies the Supervisory Authority in its application for authorisation about its intention to start cross-border activity under Article 145 or 147 of the Solvency II Directive, the notification shall contain the information described in paragraph 2.6.3.1 and the information to be notified by the Home NSA to the Host NSA in case of establishment of a branch and of starting FoS business respectively.

2.6.4 Assessment of the notification

- 2.6.4.1 The Host NSA(s) may, as soon as possible and preferably within one month upon receipt of the notification, provide the Home NSA with a feedback on the content of the notification. Such feedback can include, for example: views of the Host NSA(s) regarding the risk management measures to be put in place by the undertaking regarding risks on the distribution partners, risks on the specific products to be and additional fitness and propriety considerations.
- 2.6.4.2 In cases where an insurance undertaking applying for an authorisation has clearly indicated its intention to operate exclusively or almost exclusively in one or more Member State(s), the Host NSA(s) feedback shall provide input to facilitate the understanding of the situation and the circumstances of the undertaking by the Home NSA.

⁷ This information may be derived from the review carried out by the Home NSA concerning the notification provided by the undertaking and from the ongoing supervisory review process, including the the review of the outsourcing arrangements.

PART II AUTHORISATION

- 2.6.4.3 Where appropriate, taking into account the risk profile of the undertaking and/or the type of cross-border business EIOPA may provide the Home NSA with a feedback on the contents of the notification. In this case, such feedback shall be provided as soon as possible and preferably within 1 month upon receipt of the notification.
- 2.6.4.4 Without prejudice to its responsibility for authorisation, the Home NSA may take into account the Host NSA's and EIOPA's feedbacks when deciding to grant authorisation to the undertaking.
- 2.6.4.5 The Home NSA shall, within one month after the end of the authorisation process, inform the Host NSA(s) and EIOPA about the outcome of that process.

2.7 Acquisitions and increase of shareholdings in undertakings

Article 60 of the Solvency II Directive

- 2.7.1 The relevant Supervisory Authorities shall, as soon as possible, preferably within 2 weeks upon receipt of a request, provide each other with any information which they deem essential or relevant, be it on request of another authority, or on its own initiative, if it is proposed that an undertaking is to be acquired by any of the following:
 - a) a credit institution, insurance or reinsurance undertaking, investment firm or UCITS management company authorised in another Member State;
 - b) the parent undertaking of a credit institution, insurance or reinsurance undertaking, investment firm or UCITS management company authorised in another Member State;
 - c) a natural or legal person controlling a credit institution, insurance or reinsurance undertaking, investment firm or UCITS management company in another Member State.
- 2.7.2 The above paragraph (c) is applicable if a natural or legal person or such persons acting in concert (the proposed acquirer) have taken a decision to further increase, directly or indirectly, a qualifying holding in an insurance or reinsurance undertaking as a result of which the proportion of the voting rights or of the capital held would reach or exceed 20%, 30% or 50% or so that the insurance or reinsurance undertaking would become its subsidiary.
- 2.7.3 A decision by the Supervisory Authority that has authorised the insurance or reinsurance undertaking in which the acquisition is proposed shall indicate any

⁸ Member States need not apply the 30% threshold where, in accordance with Article 9(3)(a) of Directive 2004/109/EC, they apply a threshold of one-third.

PART II AUTHORISATION

- views or reservations expressed by the Supervisory Authority responsible for the proposed acquirer, in accordance with Article 60(2) of the Solvency II Directive.
- 2.7.4 The Supervisory Authorities shall provide upon request information on changes in qualifying holdings in the supervised undertakings to other Supervisory Authorities.

3 PART III CROSS-BORDER ACTIVITIES

3. Framework for the collaboration of Supervisory Authorities

The provisions for collaboration between Home and Host NSAs are established on the following bases:

a) Single market

The authorisation to take-up and pursue the business of insurance or reinsurance is valid for the EEA and covers the right of establishment and the freedom to provide services⁹. Once authorised by the Home NSA, insurance and reinsurance undertakings have the right to establish a branch within the territory of another Member State or may pursue their business in another Member State under the freedom to provide services. For such activities no further authorisation is needed, neither by the Home NSA nor the Host NSA. The intention to pursue insurance business in another Member State (branch operation or provision of services) has to be notified to the Home NSA. The latter communicates this to the Host NSA.

The single market approach requires that the same rules apply to all market participants and that proper consideration is given to the specificities of each insurance/reinsurance market.

b) Policyholder protection

A similar level of protection should be assured to policyholders across the EEA regardless of the location of the insurance or reinsurance undertakings' head office.

c) Supervisory cooperation and exchange of information

Information exchange and ongoing cooperation is essential for Home NSAs to perform effective supervision and for Host NSAs to address the fair treatment of policyholders. Home NSAs should make use of the knowledge of the Host NSAs about the conduct of undertakings in their territory and Host NSAs should be able to utilise the knowledge of the Home NSAs about the prudential status of the undertakings under the home country control. To this end, a set of notifications between the Home and Host NSA is put in place. Without prejudice to the Home and Host NSA responsibilities, in order to enhance the quality of information exchange, allow identification of potential risks as soon as possible and foster proactive supervision and timely supervisory actions, EIOPA should be included in all information exchanged between Home and Host NSA described in Part III.

٠

⁹ The concept of right of establishment presupposes a lasting presence in the Host Member State while provision of services is of temporary character. The temporary nature of the provision of services shall be assessed in the light of its duration, regularity, frequency and continuity, according to the case law of the Court of Justice. Additional guidance is provided by the Commission Interpretative Communication (2000/C 43/03):

3.1 Establishment, change of information and closure of a branch by a(n) (re)insurance undertaking

Articles 145-146 of the Solvency II Directive

The exchange of information under Section 3.1 shall be performed by the Home and the Host NSA by using the IT tool and the templates provided by EIOPA

Without prejudice to the responsibilities of the Home and the Host NSA, where appropriate, EIOPA shall be included for information in all formal communication between the Home and Host NSA described in Section 3.1.

3.1.1 Information to be communicated by the Home NSA to the Host NSA and EIOPA in case of insurance branch establishment

Section 3.1.1 defines the information that the Home NSA shall communicate to the Host NSA and EIOPA on the notification by an insurance undertaking to establish a branch in the Member State of the Host NSA, in particular:

- a) Paragraph 3.1.1.1 lists the information pursuant to Articles 145 and 146 of the Solvency II Directive;
- b) Paragraph 3.1.1.2 includes a non-exhaustive list of information to be communicated by the Home NSA to the Host NSA so that the latter can support the Home NSA in the supervision of the insurance undertaking's activities under the freedom of establishment;
- c) Paragraph 3.1.1.3 lists the information to be communicated by the Home NSA to the Host NSA on a non-systematic basis in order to allow the Host NSA to have a deeper understanding of the branch activity and to facilitate awareness for the ongoing supervision.
- 3.1.1.1 Where it does not oppose the establishment of a branch in another Member State, the Home NSA shall communicate the following non-exhaustive list of information to the Host NSA (communication of notification) and EIOPA:
 - a) the name and address of the head office of the insurance undertaking;
 - b) the address of the branch in the Host Member State, from which documents, including all communications to the authorised agent, may be obtained or delivered to:
 - c) the classes of insurance according to the Annexes I and II to the Solvency II Directive, into which the planned business falls;

- d) the scheme of operations setting out, at least, the following 10:
 - a. the nature of the risks or commitments which the undertaking is proposing to cover through the branch;
 - b. if available, the guiding principles as to reinsurance and to retrocession with respect to the branch operations;
 - c. if available, cost estimates for setting up the administrative services and organisation for securing business of the branch; the financial resources set aside for this purpose and, if the risks to be covered fall within class n°18 in Part A of Annex I to the Solvency II Directive, the company taking over assistance services or the resources available to the insurance undertaking to provide the promised assistance;
 - d. the organisational structure of the branch;
 - e. if available, for the first three financial years of the branch:
 - estimates of management expenses, other than installation costs. In particular, general expenses and commissions;
 - estimates of premiums or contributions and claims;
- e) the name of a person, and if available the e-mail address, who possesses sufficient powers to bind, in relation to third parties, the insurance undertaking or, in the case of Lloyd's, the underwriters concerned and to represent it or them in relations with the authorities and courts of the host Member State (the authorised agent) and description of their powers;
- f) if the insurance undertaking is to cover risks in class 10 in Part A of Annex I to the Solvency II Directive, not including carrier's liability, a declaration that the undertaking has become a member of the national bureau and the national guarantee fund of the Host Member State;
- g) in cases where the insurance undertaking intends to cover risks relating to legal expenses insurance, the option chosen from those described in Article 200 of the Solvency II Directive;

¹⁰ The items below are consistent with Article 23 of the Solvency II Directive. Some items may be influenced by the local accounting framework. When necessary the Host NSA may ask clarifications from the Home NSA with regard to the components of those items.

- h) a certificate attesting that the insurance undertaking covers the SCR and MCR as calculated in accordance with Articles 100 to 129 of the Solvency II Directive in the form provided for in Annex 1 hereto.
- 3.1.1.2 The Home NSA shall be able to make use of the knowledge of the Host NSAs about the specificities of their markets and conduct of business in their territories. In order to provide a proper context for this exchange of information (see Section 3.1.2 on the information requested from the Host NSA) the following information shall be communicated by the Home NSA together with the information listed in 3.1.1.1. to the Host NSA and EIOPA.
 - a) the legal entity identifier (LEI) of the undertaking notifying the intention to establish a branch (or if not available, the identification code used in the local market allocated by the Supervisory Authority);
 - b) if available, identification of the person(s) who effectively run the branch or are responsible for key functions for the branch¹¹, through for example the person(s) ID card, passport or nomination document(s) and if so required by the host Member State legislation, her address of residence in the host Member State;
 - c) if the undertaking belongs to a cross-border group, the name of the group supervisor and the structure of the group as included in the coordination arrangement's annex, together with the last reported group solvency position;
 - d) any available information regarding the planned distribution channel(s), relevant outsourcing contracts and partners that will be used in the Host Member State¹²;
 - e) if available the results of the assessment of the undertaking's compliance with the product oversight and governance requirements (e.g. material findings regarding design of the product, monitoring of distributors, target market identification and product testing);
 - f) a description of the relevant policyholder guarantee funds in the Home Member State, where applicable.

¹¹ Persons responsible for the activities of the branch within the undertaking (Home Member State).

¹² This information may be derived from the review carried out by the Home NSA concerning the notification provided by the undertaking including the scheme of operations. Information may also be obtained from the review of the outsourcing requirements laid down in the Solvency II Directive, its implementing measures and EIOPA's Guidelines on System of Governance.

PART III CROSS-BORDER ACTIVITIES

- 3.1.1.3 Without prejudice to the proportionality principle and the principle of risk-based supervision, the Home NSA shall consider communicating additional information to allow the Host NSA to have a deeper knowledge of the branch activity and facilitate awareness for the ongoing supervision (see Part IV). See below examples of additional information that can be provided by the Home NSA to the Host NSA and EIOPA:
 - a) a summary of the undertaking's system of governance, including the risk management system in place, which provides for the proper management of the business of the branch:
 - b) any available information resulting from the discussions with the undertaking of its business strategy and how the branch fits into that strategy.
- 3.1.1.4 The Host NSA may ask, on an ad-hoc basis, the Home NSA for information in addition to that specified in paragraphs 3.1.1.1 and 3.1.1.2 and provided under 3.1.1.3, before the undertaking establishes the branch preferably within one month upon receipt of the communication of the notification. Any such request shall be proportionate to the type of business, risks or commitments that the undertaking intends to cover in the Host NSA's territory. The Host NSA shall indicate the rationale supporting that ad-hoc request. In this case, the Home NSA shall inform the insurance undertaking of the request. The additional information requested, where possible, shall be included in an updated communication. If the insurance undertaking is not in a position to provide the additional information, the Home NSA shall inform the Host NSA.
- 3.1.1.5 The information shall be communicated by the Home NSA to the Host NSA and EIOPA, as soon as possible, and in any event within three months from receiving the complete notification from the insurance undertaking which intends to establish a branch. The Home NSA shall ensure that the Host NSA and EIOPA receives the complete notification information. Immediately upon receiving the communication from the Home NSA, the Host NSA shall acknowledge its receipt.
- 3.1.1.6 The Home NSA may where appropriate have an informal exchange of information with the Host NSA before sending the complete notification. This may allow an exchange of information before the formal commencement of branch activity.

3.1.2 Information to be provided by the Host NSA to the Home NSA and EIOPA in case of insurance branch establishment

- 3.1.2.1 The Host NSA shall communicate to the Home NSA and EIOPA:
 - a) any conditions under which, in the interest of the general good¹³, the activity must be pursued within the territory of the Host Member State or confirm that no conditions have been imposed in accordance with Article 146(3) of the Solvency II Directive. This information shall include the link to the website where the general good conditions are published;
 - b) irregularities known to the Host NSA about the planned outsourcing activities, distribution partners, claims representatives, key persons as well as any relevant information following the analysis of the notification received from the Home NSA;
 - c) if the insurance undertaking or its parent undertaking tried to establish an insurance undertaking in the Host Member State. Where an application for authorisation had been declined, the Host NSA shall provide additional information;
 - d) if a related insurance undertaking, within the meaning of Article 212 of the Solvency II Directive, of the undertaking or its parent undertaking previously established in the Host NSA had its authorisation revoked or withdrawn and the reasons supporting such a decision;
 - e) description of relevant policyholder guarantee funds in the Host Member State that would be applicable to the branch.
- 3.1.2.2 If the Host NSA considers the information contained within the communication to be incomplete, the Host NSA shall without delay inform the Home NSA. The Host NSA shall provide details of those areas where the information is considered to be incomplete and request the outstanding information.
- 3.1.2.3 All the information referred to in this Section shall be communicated as soon as possible by the Host NSA to the Home NSA and EIOPA and within two months of receipt of the communication of the notification. The Home NSA shall acknowledge receipt of all information received.
- 3.1.2.4 Without prejudice to the responsibilities of the Home and the Host NSA, where appropriate on a risk-based approach, EIOPA may provide its views to the Home NSA and/or the Host NSA(s) on the communication of notification.

-

¹³ A Supervisory Authority will not be expected to provide information on general good provisions which extend beyond those directly relating to the area of financial services.

3.1.3 Information to be provided by the Home NSA to the insurance undertaking in case of insurance branch establishment

- 3.1.3.1 Upon receipt of a notification from an insurance undertaking, the Home NSA shall assess the completeness and accuracy of the information provided. Where the information provided is incomplete or incorrect, the Home NSA shall inform the insurance undertaking in writing without delay, indicating where the information is incomplete or incorrect.
- 3.1.3.2 Within three months of receiving the complete notification from the insurance undertaking, the Home NSA shall inform the insurance undertaking concerned that the information has been communicated to the Host NSA¹⁴. Branch activity shall not commence in the Host State until the occurrence of one of the events referred to in paragraph 3.1.4. This communication shall be done preferably after receiving the acknowledgement of receipt from the Host NSA.
- 3.1.3.3 The Home NSA shall communicate to the insurance undertaking the information received from the Host NSA about the conditions under which, in the interest of the general good, the activity must be pursued within the territory of the Host State or communicate that no conditions were communicated as being imposed.
- 3.1.3.4 In the case of a refusal of the Home NSA to communicate the information referred to in Article 145(2) of the Solvency II Directive to the Host NSA, the Home NSA shall state the reasons for the refusal to the insurance undertaking within three months of receiving all the information in question.

3.1.4 Starting the activity of the branch

In accordance with Article 146(3) of the Solvency II Directive, the branch may commence activity at the earliest of either of the following:

- a) the date upon which the Home NSA has received the communication pursuant to paragraph 3.1.2.1 a); or
- b) if no such communication is received, on the expiry of the two months period from the date when the Host NSA received the notification from the Home NSA.

3.1.5 Changes to information concerning the branch

3.1.5.1 In the event of a change in any of the particulars communicated under Article 145(2) (b), (c) or (d) of the Solvency II Directive, an insurance undertaking shall give written notice of the change to the Home NSA at least one month

¹⁴ The three-month period shall be considered as having commenced on the date of receipt of the complete notification by the Home NSA.

- before making the change. This is to allow the Home NSA to fulfil their obligations under Article 146 of the Solvency II Directive.
- 3.1.5.2 If the Home NSA does not object to the proposed change(s) as notified by the insurance undertaking, it shall communicate the information, including where appropriate, an attestation that the insurance undertaking covers the SCR and the MCR. This shall be sent to the Host NSA and EIOPA as soon as possible and in any event no later than one month after it has received the information from the insurance undertaking.

3.1.6 Branch closure

- 3.1.6.1 The Home NSA shall notify the Host NSA and EIOPA if business activities will no longer be continued due to the proposed closure of the branch as soon as possible and in any case before the discontinuation of the insurance activities by the branch. Such notification shall contain, when applicable, information on the management of the run-off portfolio, for example when the run off portfolio is managed by the undertaking's head office. If applicable, the Home NSA shall ensure that the undertaking complies with the notifications requirement for FoS.
- 3.1.6.2 In the event of the closure of the branch the Home NSA shall inform, before the closure, the Host NSA how the policies underwritten by the branch will be managed. The Home NSA shall, before the closure, closely cooperate with the Host NSA in order to address any issue that policyholder protection and/or the local law applicable to the insurance contracts could raise. The management of the policies underwritten by the branch shall take into consideration the protection of policyholders established in the jurisdiction of the Host NSA.
- 3.1.6.3 The procedure mentioned in paragraph 3.1.6.1 shall apply whenever a branch no longer accepts new business and the undertaking no longer administers its portfolio of contracts via this branch.
- 3.1.6.4 After the closure the branch, as soon as possible and preferably within 1 month after the completion of the closure activities, the Home NSA shall notify the Host NSA and EIOPA. Such notification is to be performed to ensure consistency of the information available to the Host NSA and EIOPA and shall contain, when applicable, information on the management of the run-off portfolio. If applicable, the Home NSA shall ensure that the undertaking comply with the notifications requirement for FoS.

3.1.7 Establishment, change of information and closure of a branch by a reinsurance undertaking

3.1.7.1 Where a reinsurance undertaking intends to establish a branch in another Member State, the Home NSA shall, if available according to the laws of the

PART III CROSS-BORDER ACTIVITIES

Home Member State, communicate in writing the following information to the Host NSA and EIOPA:

- a) the name and address of the head office of the reinsurance undertaking;
- b) the address of the branch, which shall also be that of the authorised agent;
- c) the name and powers of the authorised agent;
- d) the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, into which the planned business falls;
- e) a certificate attesting that the reinsurance undertaking covers the SCR and MCR as calculated in accordance with Articles 100 to 129 of the Solvency II Directive, in the form provided for in Annex 1 hereto.
- 3.1.7.2 The information shall be communicated by the Home NSA to the Host NSA and EIOPA within one month after having received information from the reinsurance undertaking in respect of its intention to establish a branch in the territory of another Member State.
- 3.1.7.3 Any proposed change to the information submitted pursuant to paragraph 3.1.7.1 shall be, as far as possible, communicated by the Home NSA to the Host NSA and EIOPA as soon as possible and in any event no later than one month after having received the information from the reinsurance undertaking.
- 3.1.7.4 The Home NSA shall notify, as soon as possible and in any case before the closure, the Host NSA and EIOPA if business activities will no longer be continued due to the proposed closure of the branch.
- 3.1.7.5 After the closure the branch, as soon as possible and within one month after the completion of the closure activities, the Home NSA shall notify the Host NSA and EIOPA.

3.2 Commencing activities by way of freedom to provide services by an insurance undertaking

Articles 147 to 149 of the Solvency II Directive

The structure of 3.2 is the same as 3.1 (branches); however, it is important to take into account that in accordance with Article 148 (4) of the Solvency II Directive an insurance undertaking may start to pursue business under the freedom to provide services (FoS) business as from the date which is informed of the communication of the notification to the Host NSA by the Home NSA. As a result, in case of FoS the feedback from the Host NSA (3.2.2) shall be expected to occur after the commencing of the FoS activity by the insurance undertaking.

The exchange of information under Section 3.2 shall be performed by the Home and the Host NSA by using the IT tool and the templates provided by EIOPA

Without prejudice to the responsibilities of the Home and the Host NSA, where appropriate, EIOPA shall be included for information in all formal communication between the Home and Host NSA described in Section 3.2.

3.2.1 Information to be communicated by the Home NSA to the Host NSA and EIOPA

Articles 147-148 of the Solvency II Directive

Section 3.2.1 defines the information that the Home NSA shall communicate to the Host NSA within the territory of which the insurance undertaking intends to pursue business under FoS and EIOPA, in particular:

- a) Paragraph 3.2.1.1 lists the information pursuant to Articles 147 and 148 of the Solvency II Directive;
- b) Paragraph 3.2.1.2 includes a non-exhaustive list of information to be communicated by the Home NSA to the Host NSA so that the latter can support the Home NSA in the supervision of the insurance undertaking's activities under the FoS;
- c) Paragraph 3.2.1.3 lists the information to be provided by the Home NSA to the Host NSA on a non-systematic basis in order to allow the Host NSA to have a deeper understanding of the FoS activity and to facilitate awareness for the ongoing supervision.
- 3.2.1.1 If the Home NSA has no reasons to refuse to communicate the notification provided by the insurance undertaking to the Host NSA, it shall communicate the following non-exhaustive list of information to the Host NSA (communication of notification) and EIOPA:
 - a) the name and address of the head office of the insurance undertaking 15;
 - b) where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, from which it plans to provide services;
 - c) the classes of insurance according to the Annexes I and II to the Solvency II Directive which the insurance undertaking has been authorised to offer

¹⁵ If available it shall be indicated the email details of the head office of the insurance undertaking allowing the Host NSA to send the information referred on 3.2.3.2 by electronic means.

- d) the nature of the risks or commitments which the insurance undertaking proposes to cover in the Host Member State;
- e) if the insurance undertaking intends to cover risks in class 10 in Part A of Annex I to the Solvency II Directive not including carrier's liability:
 - i. the name and address of the claims representative as referred to in Article 18(1)(h);
 - ii. a declaration that it has become a member of the national bureau and national guarantee fund of the Host Member State;
- f) if the insurance undertaking intends to cover risks relating to legal expenses insurance, the option chosen from those described in Article 200 of the Solvency II Directive;
- g) a certificate attesting that the insurance undertaking covers the SCR and MCR calculated in accordance with Articles 100 and 129 of the Solvency II Directive, in the form provided in Annex I hereto;
- 3.2.1.2 The Home NSA shall be able to make use of the knowledge of the Host NSAs about the specificities of their markets and conduct of business in their territories. In order to provide a proper context for this exchange of information (see Section 3.2.2 on the information requested from the Host NSA) the following additional information shall be communicated by the Home NSA to the Host NSA and EIOPA:
 - a) the LEI of the undertaking intending to pursue business under FoS (or if not available, the identification code used in the local market allocated by the Supervisory Authority);
 - b) if the undertaking belongs to a cross-border group, the name of the group supervisor and the structure of the group as included in the coordination arrangement's annex, together with the last reported group solvency position;
 - c) any available information regarding expected local third or related parties involved in the underwriting activities, including but not limited to the foreseen distribution partners in the Host Member State¹⁶;

¹⁶ This information may be derived from the review carried out by the Home NSA concerning the notification provided by the undertaking and from ongoing supervisory review process (SRP). Information may also be obtained from the review of the outsourcing requirements laid down in the Solvency II Directive, its implementing measures and EIOPA's Guidelines on System of Governance.

- d) identification, through for example the person's ID card, passport or nomination document(s) and an e-mail address, of the person who is responsible within the insurance undertaking for handling of complaints in relation to the FoS activities;
- e) if available, the results of the assessment of the undertaking's compliance with the product oversight and governance requirements (e.g. material findinds regarding design of the product, monitoring of distributors, target market identification and product testing);
- f) description of the relevant policyholder guarantee funds in the Home Member State, where applicable.
- 3.2.1.3 Where the undertaking intends to operate exclusively, or almost exclusively, in other EEA Member State(s) on FoS basis and without prejudice to the proportionality principle and the principle of risk-based supervision, the Home NSA shall consider communicating additional information to allow the Host NSA to have a deeper knowledge of the FoS activity and facilitate awareness for ongoing supervision (see Part IV). See below examples of additional information that could be provided by the Home NSA to the Host NSA and EIOPA:
 - identification, through for example the persons(s) ID card, passport or nomination document(s), of the personsn who effectively run the undertaking or are responsible for the key functions. This information shall be communicated in all applicable cases;
 - b) a summary of the undertaking's system of governance, including the risk management system in place;
 - c) any available information resulting from the discussions with the undertaking of its business strategy and how the FoS fits into that strategy.
- 3.2.1.4 The Host NSA may ask for ad-hoc information from the Home NSA in addition to that specified in paragraphs 3.2.1.1 and 3.2.1.2 and provided under 3.2.1.3¹⁷ Any such request shall be proportionate to the type of business, risks or commitments that the undertaking intends to cover in the Host NSA's territory. The Host NSA shall indicate the rationale supporting that ad-hoc request. The Home NSA shall inform the insurance undertaking of the request. If the

¹⁷ For instance if the Host NSA has doubts as to the precise conditions under which the activity is to be pursued, in which case it may ask the Home NSA about the specific resources which the insurance undertaking proposes to use in marketing its products in the host MemberState.

insurance undertaking is not in a position to provide the additional information, the Home NSA shall inform the Host NSA.

- 3.2.1.5 The information shall be communicated by the Home NSA to the Host NSA and EIOPA, as soon as possible, and in any event within one month from receiving the complete notification from the insurance undertaking which intends to carry on business by way of FOS in the territory of another Member State. The Home NSA shall ensure that the Host NSA receives the complete notification. Immediately upon receiving the communication of notification from the Home NSA, the Host NSA shall acknowledge its receipt.
- 3.2.1.6 Where the undertaking intends to operate exclusively, or almost exclusively, in other EEA Member State(s) on FoS basis, the Home NSA shall evaluate whether to have an informal exchange of information with the Host NSA before sending the complete notification. This should allow an exchange of information before the formal start of the activity by FoS.
- 3.2.1.7 In the other cases, i.e. where the undertaking does not intend to operate exclusively, or almost exclusively, in other EEA Member States on FoS basis, the Home NSA may, on a case by case approach, have an informal exchange of information with the Host NSA before sending the complete notification.

3.2.2 Information to be provided by the Host NSA to the Home NSA and EIOPA

- 3.2.2.1 The Host NSA shall communicate to the Home NSA and EIOPA:
 - a) any conditions under which, in the interest of the general good¹⁸, the
 activity must be pursued within the territory of the Host Member State or
 confirm that no conditions have been imposed. This information shall
 include the link to the website where the general good conditions are
 published;
 - b) any irregularities known to the Host NSA, about the local third or related parties involved in the underwriting activities in the Host Member State, about claims representatives, about key persons as well as any relevant information following the analysis of the notification received from the Home NSA;
 - c) if the insurance undertaking or its parent undertaking tried to establish an insurance undertaking in the Host Member State. Where an application for

¹⁸ A Supervisory Authority will not be expected to provide information on general good provisions which extend beyond those directly relating to the area of financial services.

authorisation had been declined, the Host NSA shall provide additional information:

- d) if a related insurance undertaking of the undertaking or its parent, within the meaning of Article 212 of the Solvency II Directive, previously established in the Host NSA had its authorisation revoked or withdrawn and the reasons supporting such a decision;
- e) description of relevant policyholder guarantee funds in the Host Member State that would be applicable to the FoS.
- 3.2.2.2 If the Host NSA considers the information contained within the communication to be incomplete, the Host NSA shall without delay inform the Home NSA. The Host NSA shall provide details of those areas where the information is considered to be incomplete and request the outstanding information.
- 3.2.2.3 All the information referred in this Section shall be communicated as soon as possible by the Host NSA to the Home NSA and EIOPA. The Home NSA shall acknowledge receipt of all information received.
- 3.2.2.4 Without prejudice to the responsibilities of the Home and the Host NSA, were appropriate on a risk-based approach, EIOPA may provide its views to the Home NSA and/or the Host NSA(s) on the communication of notification.

3.2.3 Information to be provided by the Home and Host NSAs to the insurance undertaking

3.2.3.1 By the Home NSA to the insurance undertaking

- 3.2.3.1.1 Upon receipt of a notification from an insurance undertaking, the Home NSA shall assess the completeness and accuracy of the information provided. Where the information provided is incomplete or incorrect, the Home NSA shall inform the insurance undertaking in writing without delay, indicating in which respect the information is incomplete or incorrect.
- 3.2.3.1.2 The Home NSA shall inform the insurance undertaking concerned that the notification information has been sent to the Host State (see 3.2.4).
- 3.2.3.1.3 In case of refusal of the Home NSA to communicate the information referred to in Article 148(1) of the Solvency II Directive to the Host NSA, the Home NSA shall state the reasons for the refusal to the insurance undertaking within one month of receiving all the information in question.

3.2.3.2 By the Host NSA to the insurance undertaking

The Host NSA shall as soon as possible after the receipt of the communication of the notification, communicate in writing to the head office of the insurance undertaking any conditions, under which, in the interests of the general good, the activity must be pursued within the territory of the Host State or communicate that no conditions are imposed.

3.2.4 Starting of activity by the insurance undertaking

In accordance with Article 148(4) of the Solvency II Directive, the insurance undertaking may commence activity by way of FoS when it is informed by the Home NSA that the communication of notification provided for in Article 147 to the Host NSA has been made.

3.2.5 Changes to information

- 3.2.5.1 In the event of a change in any of the particulars communicated under Articles 147 of the Solvency II Directive, the insurance undertaking shall give written notice of the change to the Home NSA.
- 3.2.5.2 The Home NSA shall communicate the information to the Host NSA and EIOPA as soon as possible, but in any event, no later than one month after it has received the information from the insurance undertaking.

3.2.6 Cessation of Activities

- 3.2.6.1 The Home NSA shall notify the Host NSA and EIOPA, if the information is available, if business activities will no longer be provided by FoS in the Host Member State, as soon as possible and before the discontinuation of the insurance activities. This shall include stop of writing new business, termination of the existing policies and settlement of the outstanding claims.
- 3.2.6.2 As soon as possible and preferably within one month after the effective discontinuation of the insurance activities, the Home NSA shall notify the Host NSA and EIOPA. Such notification is to be performed to ensure consistency of the information available to the Host NSA and EIOPA and shall contain, where applicable, information on the management of the run-off portfolio.

3.3 Other provisions common to branches and freedom to provide services

3.3.1 Language, communication means, contact points and lists

3.3.1.1 All information exchanged between Supervisory Authorities in respect of notifications and changes to cross-border activity, shall be written in a language which is acceptable by the Supervisory Authorities in accordance with paragraph 3.3.1.4 b) where the Supervisory Authorities have not agreed bilaterally on other language for the exchange. The Supervisory Authorities shall make every

PART III CROSS-BORDER ACTIVITIES

effort to be as flexible as possible in respect of languages in which they accept notifications.

- 3.3.1.2 Each Supervisory Authority shall transmit the required information by electronic means (see contact points in paragraph 3.3.1.4 a)) where possible and in a format acceptable to the relevant Supervisory Authorities. The Supervisory Authorities shall agree on any applicable requirements (e.g. requirement on electronic signatures of authorised persons) and the security of data transmission.
- 3.3.1.3 Each Supervisory Authority shall establish a contact point or set up a general email account for all questions, requests and problems arising out from the notifications and changes to cross-border business activities. This contact point shall be used for all queries regarding cross-border business in order to avoid unintended delays.
- 3.3.1.4 Each Supervisory Authority shall keep the following information up to date on EIOPA's extranet and on the IT tool provided by EIOPA:
 - a) details of the contact point for all questions, requests and problems arising from the notification and changes to cross-border business activities, including name, email address and phone number. The contact point may be a general e-mail account, to which information shall be sent;
 - b) the language(s) in which the Supervisory Authority accepts documents in respect of the notification of cross-border activities.
- 3.3.1.5 The Host NSAs shall upload on their website a list of all insurance undertakings which have notified their intention to establish a branch or provide business on a FoS basis in its territory once the information is communicated by the Home NSA. The information shall be updated as necessary by the Home NSA.
- 3.3.1.6 The Supervisory Authorities shall maintain a data storage system that allows the extraction of information on an individual and aggregated basis. The data stored shall include at least the following:
 - a) As a Home NSA, the data provided under paragraphs 3.1.1.1 a) to c), d) (a.), e) and f) and 3.2.1.1 a) to f) to the Host NSAs and the data received from the Host NSAs pursuant to paragraphs 3.1.2.1 and 3.2.2.1;
 - b) As a Host NSA, the data received from the Home NSAs as per point a) and the data sent pursuant to paragraphs 3.1.2.1 and 3.2.2.1.

The Home and Host NSAs shall be able to analyse this data electronically (e.g. being able to extract a list of notifications per Member State, per insurance

- undertaking and per period and on an individual and aggregated basis). The referred data storage system shall apply to the new branch and FoS notifications.
- Without prejudice to the data storage systems of the NSAs, the IT tool provided by EIOPA funtions as the data storage system referred to in paragraph 3.3.1.6.
- 3.3.1.7 Each Supervisory Authority shall publish an up-to-date list of the general good conditions of the respective jurisdiction on its website or communicate that no conditions are imposed, in its own language(s) and/or in English.
- 3.3.1.8 The Supervisory Authorities shall provide EIOPA with the links to where the general good conditions are published on the relevant websites in their jurisdiction. EIOPA shall publish these links on the Public Area of the EIOPA website.

3.3.2 Submission of policy conditions to the Host NSA (health insurance as an alternative to social security)

Article 206 (1) of the Solvency II Directive

- 3.3.2.1 The Host NSA shall inform the relevant Supervisory Authorities of instances of contracts covering the risks under class 2 in Part A of Annex I of the Solvency II Directive, where these contracts may serve as a partial or complete alternative to health cover provided by the statutory social security system, and the general and specific conditions of such policies must be provided to the Host NSA before they are applied. This information shall be updated as soon as existing legislation is changed or new legislation is introduced. This information shall include a link to the relevant website of the Host NSA.
- 3.3.2.2 The Home NSAs, who receives the information mentioned above, shall transmit it to the insurance undertakings wishing to establish a branch or take up FoS in the Member State concerned. The Home NSA shall advise that such contracts may not be concluded before the general and special conditions of that insurance are communicated to the Host NSA in the language(s) specified by the Host Member State's national law.
- 3.3.2.3 The Supervisory Authorities shall provide EIOPA with the link to the websites where the insurance classes and/or insurance products are listed, for which policy conditions have to be presented to the Host NSA. The aforementioned information shall be updated as necessary. EIOPA shall publish the relevant links on the Public Area of the EIOPA website.

3.3.3 Representative for the handling of claims (motor vehicle liability)

3.3.3.1 In the area of motor vehicle liability insurance, and preferably prior to the notification from the Home NSA to the Host NSA, Supervisory Authorities shall co-operate with the Home NSA in case the Home NSA performs a check of the

PART III CROSS-BORDER ACTIVITIES

reliability and the professional qualification of the representatives mentioned in Article 152 of the Solvency II Directive, including the claims representative appointed pursuant to Article 4 of Directive 2000/26/EC.

3.3.3.2 The requested Supervisory Authority shall, as far as possible, endeavour to promptly supply the information requested by the Home NSA or confirm that they have no such information.

Articles 29, 30, 33, 152a, 152b, 155 and 158 of the Solvency II Directive

The on-going collaboration between Home and Host NSAs is crucial to ensure the proper functioning of the single market by maintaining a level playing field for undertakings and ensuring the same level of consumer protection for policyholders.

In order to ensure that the Home and the Host NSAs are equipped with the most updated supervisory information, Part IV of this Decision defines:

- a) areas of on-going cooperation between Home and Host NSAs;
- b) the process to perform the notifications in accordance with Article 152a of Solvency II Directive;
- c) the process to exchange information as laid down in Articles 30, 33, 155 and 158 of Solvency II Directive;
- d) the information to be exchanged by NSAs in case of supervisory measures taken on cross-border undertakings;
- e) conditions for participation in the meetings of a college of supervisors for Host NCAs and the conditions to set up a collaboration platform in accordance with Article 152b of Solvency II Directive
- f) the process to exchange information on portfolio transfer as laid down in Articles 39 and 164 of Solvency II Directive.

4.1 Continuing cooperation between Home and Host NSAs (branches and FoS)

- 4.1.1.1 The financial supervision of the activities of insurance and reinsurance undertakings, both off-site and on-site, shall be the sole responsibility of the Home NSA in accordance with Article 30 of the Solvency II Directive. It shall encompass the supervision of the business pursued either through branches or under FoS but without prejudice to the powers of the Host NSA as recognised by Article 155 of that Directive.
- 4.1.1.2 The Home NSA shall consider increasing its cooperation with the Host NSAs to understand, within its continuous supervisory review process whether the insurance and reinsurance undertaking has a clear understanding of the risks that it faces, or may face, in the Host territories; and which specific related risk management tools and internal controls are in place, having regard to the

proportionality principle and the risk-based approach to supervision. With the Host NSA's local knowledge in mind, as regards actual and potential risks, there shall be a particular focus in terms of cooperation on the following risk areas:

- a) system of governance including the ability of the head office management to understand the cross-border market specificities, related risk management tools and internal controls in place. In regards to the risk management system special attention shall be paid to underwriting, pricing and reserving;
- b) outsourcing contracts and distributions partners;
- c) claims handling;
- d) compliance;
- e) consumer protection.

4.2 Exchange of information

4.2.1 Notifications from the Home NSA

- 4.2.1.1 In accordance with Article 152a(2) of Solvency II Directive, the Home NSA shall notify EIOPA and the Host NSA(s) when an undertaking carrying out cross-border activity under FoE or FoS shows deteriorating financial conditions or other emerging risks which may have cross border effects.
- 4.2.1.2 In order to achieve supervisory convergence in the notification practices, a set of thresholds and criteria have been defined.
- 4.2.1.3 The notifications described in Section 4.2.1 shall be performed by the Supervisory Authorities by using the IT tool and the templates provided by EIOPA

Materiality thresholds

- 4.2.1.4 The Home NSA shall make the notifications, if at least one of the following conditions is met:
 - a) the undertaking carrying out cross-border activity under FoE or FoS has annual gross written premiums in the Host Member State(s) higher than EUR 250,000 according to the lastest information at the Home NSA disposal or
 - b) the undertaking carrying out cross-border activity under FoE or FoS has reported in its balance sheet technical provisions related to risks

underwritten in the Host Member State higher than EUR 1,000,000 according to the lastest information at the Home NSA disposal.

Without prejudice to the above, in case the undertaking carrying out cross-border activity is a captive (re)insurance undertaking, the Home NSA may not perform the notification if the following conditions are met:

- in relation to the insurance obligations of the captive (re)insurance undertaking, all insured and beneficiaries are legal entities of the group or natural persons eligible to be covered under the group insurance policies of which the captive (re)insurance undertaking is part, as long as the business covering natural persons eligible to be covered under the group insurance policies remains immaterial;
- in relation to the reinsurance obligations of the captive (re)insurance undertaking, all insured and beneficiaries of the insurance contracts underlying the reinsurance obligations are legal entities of the group of which the captive (re)insurance undertaking is part;
- the insurance obligations and the insurance contracts underlying the reinsurance obligations of the captive (re)insurance undertaking do not relate to any compulsory third party liability insurance.

The Home NSA shall consider the materiality thresholds described above as 'de minimis' thresholds. The Home NSA may perform the notifications under Article 152a(2) of the Solvency II Directive applying lower thresholds, where appropriate.

Thresholds and criteria in case of deterioration of financial conditions

- 4.2.1.5 The Home NSA shall consider if an undertaking is showing deteriorating financial conditions by the application of the following cap and floor limits and supervisory findings described at paragraph 4.2.1.7.
 - a) The events included in the floor limits shall be considered by the Home NSA as a hard threshold. Therefore, when one of the following floor limits is reached, the Home NSA shall make a notification:
 - a. the Home NSA has observed a MCR/SCR breach¹⁹;

¹⁹ Please refer to the EIOPA Statement on supervisory practices and expectations in case of breach of the Solvency Capital Requirement for the approach to define the date of observation of non-compliance by an undertaking with the SCR.

- b. the Home NSA has observed a near breach of the SCR²⁰. In this case, the notification shall be performed if the undertaking does not increase the SCR coverage as requested by the Home NSA. If the undertaking increases the SCR coverage, the Home NSA shall evaluate on a case-by-case basis whether to perform the notification in the same way as defined for the cap limits;
- c. the Home NSA intends to take one of the following supervisory measures:
 - request a recovery plan or activate a recovery plan already submitted to the Home NSA, if applicable;
 - ii. request to materially increase the technical provisions. In this case, the notification shall be performed if the undertaking does not increase the technical provisions as requested by the Home NSA. If the undertaking increases the technical provisions, the Home NSA shall evaluate on a case-by-case basis whether to perform the notification in the same way as foreseen for the cap limits;
 - iii. request to temporarily suspend or limit the right of policyholders to surrender;
 - iv. restriction or prohibition of free disposal of assets.
- b) The events included in the cap limits shall be considered as soft threshold.. When one of the following cap limits is reached, the Home NSA shall evaluate, where appropriate, whether to make a notification:
 - a. the Home NSA has observed/been informed that the SCR/MCR ratio of the undertaking breaches the SCR/MCR ratio thresholds according to the Risk Appetite Framework of the undertaking;
 - the Home NSA has observed/been informed that the SCR/MCR ratio of the undertaking breaches the SCR/MCR ratio threshold considered by the NSA to assess situations of near non-compliance with SCR/MCR, when applicable;
 - c. the Home NSA has observed/been informed that the undertaking has materially breached one or more of other key operational

39/101

²⁰ A near non-compliance of SCR refers to the situation where the solvency position of an insurer starts to deteriorate and where it is likely that it will continue to deteriorate and fall below the SCR if no remedial action is taken.

- triggers set by the undertaking itself (e.g. capital buffer, liquidity triggers, combined ratio, surrenders ratio etc.);
- d. the Home NSA has observed/been informed of a significant drop of the SCR ratio of the undertaking (e.g. -30pp or -1/5 compared to previous period) and/or a significant increase of the volatility of the SCR ratio (e.g. changes +/- 20pp or +/-1/7 compared to the previous period) taking into account the business model of the undertaking, and the historical volatility of the SCR ratio;
- e. the Home NSA has observed a significant rating downgrade of the undertaking (e.g. . -1 notch in the credit quality step) taking into account the previous position of the undertaking.

Thresholds and criteria in case of other emerging risks with cross-border effects

- 4.2.1.6 The Home NSA shall identify an undertaking posing emerging risks which may have a cross-border effects by the application of the following limits and supervisory findings described at paragraph 4.2.1.7.
 - a) The events listed below shall be considered as soft threshold. When one of the following limits is reached, the Home NSA shall evaluate, where appropriate, whether to make a notification.
 - a. the Home NSA has classified the undertaking as "intensive monitoring/high risk" as result of its own risk assessment framework in accordance with Section III of EIOPA Guidelines on supervisory review process (EIOPA- BoS-14/179);
 - b. the Home NSA has taken enforcement measures/actions against person(s) who effectively run the undertaking or hold other key functions²¹, or against its shareholders and members with qualifying holdings;
 - c. the Home NSA has imposed any administrative sanctions or other measures against the undertaking in accordance with Solvency II Directive;

²¹ For example in case the Home NSA has the intention to request the undertaking to replace a Board member, key function holder, or person who effectively run the undertaking because they are unfit or because convicted of financial crime.

- d. the Home NSA has imposed administrative sanctions or other measures in accordance with Chapter VII of Directive (EU) 2016/97 'on insurance distribution' towards the undertaking or its intermediaries:
- e. the Home NSA has imposed administrative sanctions or other measures in accordance with Chapter V of Regulation (EU) 1286/2014 'on key information documents for packaged retail and insurance-based investment products (PRIIPs)' towards the undertaking or its intermediaries;

Qualitative events (Supervisory findings)

- 4.2.1.7 When the Home NSA performs on-site inpections or off-site activities and it identifies one or more supervisory findings, the Home NSA may consider, where appropriate, to evaluate whether the undertaking is at risk of deterioration of financial conditions (as described in paragraph 4.2.1.5) or other emerging risks which may have a cross-border effect (as described in paragraph 4.2.1.6). The following non-exhaustive list of supervisory findings shall be considered:
 - a) findings with a material impacts on the undertakings technical provisions, own funds or SCR:
 - b) findings related to the external auditor's opinion (if any) on the undertaking Solvency II balance sheet (or Solvency II reporting) or information shared by the statutory auditors, in the context of the dialogue with NSAs²², that is expected to impact the solvency position of the concerned undertaking;
 - c) findings on an unexpected increase/decrease in the undertaking key risk or performance indicators (for example, an unexpected increase of life contracts surrenders which have a material impact on the balance sheet of the undertaking).
 - d) findings, on the undertaking system of governance, with material impact on the continuous compliance of the undertaking with the applicable regulatory requirements;
 - e) findings of an unexpected loss of AMSB member or key staff who have not been replaced;

²² Please refer to EIOPA's Guidelines on facilitating an effective dialogue between competent authorities supervising insurance undertakings and statutory auditor(s) and the audit firm(s) carrying out the statutory audit of those undertakings.

- f) findings on the portfolio of the undertaking in the Host member state(s)
 (e.g. underserving, material flaws in the underwriting, claims and
 complaints processes, non-compliance with Prudent Person Principle;
- g) findings related to the data quality of the reporting materially impairing the reliability of the undertaking's financial and prudential reporting;
- h) findings in relation to the undertaking's compliance with the product oversight and governance requirements which concluded that the products manufactured by the undertaking are particularly complex or which carry significant risk for the consumer of the host Member State;
- i) the Home NSA has detected the occurrence of operational risk events on the undertaking (e.g. data breach, IT risk event, frauds, sanctions on noninsurance related regulations – e.g. privacy, AML/CTF).

Content of the notifications

- 4.2.1.8 The notifications performed by the Home NSA to EIOPA and the Host NSA(s) shall contain the following information:
 - a) the name and address of the head office of the undertaking;
 - b) a description of the event(s) which have occurred to the undertaking and a summary of the first Home NSA assessment, including where appropriate detailed documentation to understand the context;
 - c) the supervisory measures requested by the Home NSA from the undertaking and the timeline for such measures to be completed;
 - d) when available, an overview of the undertaking financial and solvency position, including where appropriate, a forward looking projection of the impacts of the supervisory measures requested by the Home NSA;
 - e) where appropriate, the Home NSA's reasoned request to the Host NSA(s) to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform;
 - f) the Home NSA's request of information to the Host NSA(s) needed to perform its supervisory activities on the undertaking;
 - g) other additional information as deemed necessary by the Home NSA.

Process of the notifications

- 4.2.1.9 The Home NSA shall notify the Host NSA and EIOPA when the floor limits are reached or the Home NSA deems it appropriate as soon as possible and preferably within one month.
- 4.2.1.10 Immediately upon receiving the notification from the Home NSA, the Host NSA(s) and EIOPA shall acknowledge their receipt.
- 4.2.1.11 Upon receipt of the notification, the Host NSA and EIOPA shall assess the content of the notification. Where appropriate, the Home NSA and EIOPA may request the Home NSA to provide further information. The Home NSA shall respond as soon as possible and preferably within one month. If the Home NSA requests information from the Host NSA(s), the same rule shall apply.
- 4.2.1.12 The Home NSA, where appropriate and as the information becomes available, shall within one month, provide updates to the Host NSA(s) and EIOPA of the financial and solvency position of the undertaking and progress regarding the implementation of the required supervisory measures. The Home NSA shall also inform the Host NSA(s) and EIOPA when the undertaking resolved its difficulties.

4.2.2 Other exchange of information from the Home NSA

- 4.2.2.1 In order to allow and facilitate the exercise of the relevant supervisory tasks the Home NSA shall provide the affected Host NSA(s) and EIOPA with the following information in a timely manner:
 - a) changes to the assessment of the suitability of shareholders and members with qualifying holdings, as well as to the assessment of the fitness and propriety of all persons who effectively run the undertaking or hold other key functions, which are connected to other member States, in line with Section 2.4;
 - b) details on undertakings showing deteriorating financial conditions or other emerging risks not reported due the application of the materiality thresholds described at paragraph 4.2.1. The Home NSA shall share with the Host NSA and EIOPA details on those undertakings and on the supervisory measures taken on them;
 - c) in case of cessation of the activities (branch or FoS), details as to how the cross-border policies covering risks or commitments situated in the territory of the Host Member State are managed (see 3.1.6, 3.1.7.4 and 3.2.6);

- d) follow-up on investigation requests from the Host NSA following paragraph 4.2.4.7 and measures taken following 4.2.2.2 and 4.2.4.13²³.
- 4.2.2.2 Where the Home NSA is informed in accordance with paragraph 4.2.3 about an insurance or reinsurance undertaking's failure to take actions to comply with the legal provisions applicable to it in the Host NSA, it shall, at the earliest opportunity, take all appropriate actions to ensure that the insurance undertaking remedies the irregular situation. The Home NSA shall inform the Host NSA about the actions taken by the Home NSA and the insurance undertaking. Any deviation from measures proposed by the Host NSA shall be properly explained to the Host NSA. Furthermore, the Home NSA shall provide information in cases where the Host NSA has already raised concerns.
- 4.2.2.3 Where in accordance with Article 33 of the Solvency II Directive, the Home NSA decides to carry out on-site inspections in a branch of an insurance or reinsurance undertaking situated in another Member State, it shall communicate it to the Host NSA in advance, preferably 4 weeks before the on-site inspection date, indicating the:
 - name and position of the persons responsible for the investigation;
 - dates planned for the action;
 - reason(s) for the investigation; and
 - programme for the proposed investigation.
- 4.2.2.4 On-site inspections of branches of an insurance or reinsurance undertaking shall enable the Supervisory Authorities to assess the situation of each establishment and the standard of its business activities. To this end, the Home NSA shall ask the insurance or reinsurance undertaking to place at its disposal, locally, any documents, books, registers, contracts, statements of claims, accounting documents, etc. which may be required, and personnel qualified to provide it with the information required.
- 4.2.2.5 Paragraph 4.2.2.3 shall not restrict the power of the Home NSA to extend its investigations beyond the initial programme. In such case, the Home NSA shall inform the Host NSA of such an extension.
- 4.2.2.6 The Host NSA may take part in the on-site inspection in accordance with Article 33 of the Solvency II Directive. If it chooses to do so, the Home NSA shall be informed without delay by the Host NSA which shall indicate the name and

²³ If the Home NSA is unable to reply within the time indicated in the request it shall inform the other Supervisory Authority of the time by which it will provide feedback. If the information is not available, it shall inform the other Supervisory Authority accordingly.

position of the persons who participate in the inspection. In accordance with Article 21(1) of Regulation (EU) No 1094/2010, EIOPA may participate in onsite inspections where they are carried out jointly by two or more Supervisory Authorities.

- 4.2.2.7 After concluding the on-site inspection, the Home NSA shall communicate the observations from the investigation, and any consequences that may arise, to the Host NSA.
- 4.2.2.8 If, at the express request of the Home NSA the Host NSA carries out an on-site inspection alone, on behalf of the Home NSA, the Home NSA shall place at the disposal of the persons empowered to carry out that investigation any accounts, documents and information which they may need in the performance of their duties.

4.2.3 Notification from the Host NSA

- 4.2.3.1 Pursuant to Article 152a(2) of Solvency II Directive, the Host NSA, where appropriate, may notify the Home NSA when it has serious and reasoned concerns with regard to consumer protection regarding an undertaking carrying out cross-border activity under FoE or FoS in its Member State. The Host NSA, where appropriate, may also notify EIOPA.
- 4.2.3.2 In order to achieve supervisory convergence in the notification practices referred to in paragraph 4.2.3.1, a set of thresholds and criteria have been defined.
- 4.2.3.3 The notifications described in Section 4.2.3 shall be performed by the Supervisory Authorities by using the IT tool and the templates provided by EIOPA.

Thresholds and criteria in case of serious and reasoned concerns with regard to consumer protection

- 4.2.3.4 The Host NSA shall consider the following list of events to identify undertakings potentially posing serious concerns with regard to consumer protection.
 - a) the Host NSA has received or has been notified by the relevant statutory authority a considerable number of complaints given the relevant market share of the undertaking;
 - b) the Host NSA has imposed administrative sanctions or other measures in accordance with Chapter V of Regulation (EU) 1286/2014 'on key information documents for packaged retail and insurance-based investment products (PRIIPs)' towards the undertaking or its intermediaries:

- the Host NSA has detected a non-compliance of the undertaking with the legal provisions of the Host Member State with impacts on consumers;
- d) the Host NSA has other, specific concerns of material relevance to the Home NSA or to EIOPA on the undertaking or its intermediaries conduct of business (e.g. failure/crime of business partner/distributor, frauds in the product/LoB, delays in paying claims, whistleblowing, etc.).
- e) the Host NSA has detected a non-compliance of material relevance with rules relating to activities of insurance distribution in accordance with Commission Delegated Regulation (EU) 2017/2358 supplementing Directive (EU) 2016/97;
- f) the Host NSA has taken emergency measures to prevent irregularities committed by the undertaking on its territory. The measures may, in particular, include a prohibition on the undertaking from continuing to conclude new contracts in the territory of the Member State concerned, or any other measure provided for by national legislation.

Content of the notifications

- 4.2.3.5 The notifications performed by the Host NSA to EIOPA and the Home NSA(s) for the purpose of Section 4.2.3 shall contain the following information:
 - a) the name and address of the head office of the insurance undertaking;
 - b) where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), or of the branch situated in the Host Member State:
 - c) a description of the event(s) occurred to the undertaking and a summary of the Host NSA assessment of the situation, including where appropriate any documentation that can clarify the issue;
 - d) the supervisory measures requested by the Host NSA from the undertaking including their timeline of application;
 - e) where appropriate, the Host NSA's reasoned request to the Home NSA to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform;
 - f) the Host NSA's request of information to the Home NSA needed to perform its supervisory activities on the undertaking;
 - g) other additional information as deemed necessary and proportionate by the Host NSA.

Process of the notifications

- 4.2.3.6 The Host NSA may notify the Home NSA for the purpose of Section 4.2.3 as soon as possible and preferably within one month after the Host NSA identified an undertaking posing serious and reasoned concerns with regard to consumer protection. When appropriate, the Host NSA may include other Host NSAs in the notification.
- 4.2.3.7 Where the Host NSA decides to notify a cross-border issue related to consumer protection to Home NSA, EIOPA and the other Host NSAs (when applicable), the Home NSA, EIOPA and the other Host NSA (when applicable) shall acknowledge their receipt of the notification immediately.
- 4.2.3.8 Upon receipt of the notification, the Home NSA and EIOPA shall assess the content of the notification. When appropriate, the Home NSA and EIOPA may request the Host NSA to provide further information. The Host NSA shall respond as soon as possible and preferably within one month. If the Host NSA requests information from the Home NSA, the same rule shall apply..
- 4.2.3.9 The Host NSA, where appropriate and as the information becomes available, shall inform in a timely manner the Home NSA, the other Host NSA(s) and EIOPA of developments in relation to the matters raised. The Host NSA shall also inform the Home NSA when the undertaking resolved its difficulties.

4.2.4 Other exchange of information from the Host NSA

- 4.2.4.1 The Host NSA shall inform in accordance with Article 30(3) of the Solvency II Directive the Home NSA if it has reasons to consider that the activities of an insurance or reinsurance undertaking might affect its financial soundness. The following are examples of information that shall be communicated by the Host NSA to the Home NSA at the earliest opportunity:
 - a) situations that can influence the Home NSA's assessment of the suitability of shareholders and members with qualifying holdings as well as, of the fitness and propriety of all persons who effectively run the undertaking or hold other key functions, in line with Section 2.4;
 - b) vulnerabilities discovered during the monitoring of compliance with the legal provisions of the Host Member State (4.2.4.2 and 4.2.4.4) through the notification described in Section 4.2.3 in case they raise concerns to consumer protection;
 - c) vulnerabilities discovered following conduct of business supervision through the notification described in Section 4.2.3 in case they raise concerns to consumer protection;

- d) potential irregularities including cases where the undertaking is pursuing an activity that has not been notified or has violated the legal provisions in force in the host Member State (4.2.4.11);
- e) undertakings' lack of action and proposed supervisory measures (4.2.4.12) through the notification described in Section 4.2.3 in case they raise concerns to consumer protection;
- f) measures taken by the Host NSA (4.2.4.14) through the notification described in Section 4.2.3 in case they generate concerns to consumer protection;
- g) other adverse developments such as significant increase of policyholders' complaints together with the complaints' topic(s), change of standards of the conduct of business, unsatisfactory commercial practices and any other issue that can significantly influence the Home NSA's assessment of the undertaking.
- 4.2.4.2 In order to investigate compliance with the legal provisions of the Host Member State also applicable to cross-border activities, any insurance or reinsurance undertaking operating through a branch or FoS, shall be requested to communicate to the Host NSA any documents the provision of which would have been compulsory if requested from an undertaking with its head office in that Host Member State.
- 4.2.4.3 Refusal to communicate such documents shall be treated as an irregularity within the meaning of Article 155 of the Solvency II Directive.
- 4.2.4.4 Where in order to investigate compliance with the legal provisions applicable in the Host Member State it is considered necessary to carry out an on-site inspection at the branch or an on-site inspection of an outsourced services provider (e.g. a managing general agent) where business is being carried out on an FoS basis, the Host NSA may carry out such an investigation on its own initiative.
- 4.2.4.5 In case of an investigation referred to in paragraph 4.2.4.4, the Host NSA shall inform the Home NSA of its decision as soon as possible, preferably 4 weeks before the on-site inspection date, communicating the
 - name and position of the persons responsible for the investigation;
 - dates planned for the action;
 - reason for the inspection; and
 - proposed programme.

- 4.2.4.6 The Home NSA may participate in the on-site inspection. If it chooses to do so, the Host NSA shall be informed without delay by the other Supervisory Authority which shall indicate the name and position of the persons who will participate in the inspection. EIOPA may participate in on-site inspections where they are carried out jointly by two or more Supervisory Authorities.
- 4.2.4.7 When, for the purposes of monitoring compliance with the legal provisions applying to a branch or to operations conducted within its territory by way of FoS, an on-site inspection at the head office of the insurance or reinsurance undertaking is considered necessary, the Host NSA shall inform the Home NSA. The Home NSA may carry out the investigation and the Host NSA can participate.
- 4.2.4.8 Where the Home NSA would experience difficulties exercising its right to carry out an on-site inspection at the branch in the Host Member State or where the Host NSA would experience difficulties exercising its right to participate in the on-site inspection of such a branch, the Supervisory Authorities may refer the matter to EIOPA for mediation.
- 4.2.4.9 The Supervisory Authority carrying out the inspection shall inform the other Supervisory Authority about the observations from the investigation.
- 4.2.4.10 Where the Host NSA ascertains that an insurance or reinsurance undertaking with a branch or pursuing business under FoS is not complying with the legal provisions applicable in the Host State, it shall require the insurance or reinsurance to remedy such irregularity.
- 4.2.4.11 The Host NSA shall communicate in writing to the head office of the insurance or reinsurance undertaking, stating the type of infringement(s) observed and the measures to be taken. A copy of this communication shall also be sent to the Home NSA. Where the operations are engaged in by a branch, a copy of this communication shall also be sent to the authorised agent of the Branch.
- 4.2.4.12 If the insurance or reinsurance undertaking concerned fails to take the necessary action, the Host NSA shall inform the Home NSA and submit to the latter all relevant information, including an assessment of the situation and the supervisory measures proposed. The Home NSA shall acknowledge receipt of this information.
- 4.2.4.13 Where despite the measures taken by the Home NSA or because those measures prove to be inadequate or are lacking in the Home Member State, and the insurance or reinsurance undertaking persists in violating the legal provisions in force in the Host NSA, the Host NSA may:

- a) take appropriate measures, after informing the Home NSA, to prevent or penalise further irregularities, including, in so far as is strictly necessary, preventing the insurance or reinsurance undertaking from continuing to conclude new (re)insurance contracts with the territory of the Host Member State;
- b) refer the matter to EIOPA for binding mediation.
- 4.2.4.14 Notwithstanding the procedure defined above, in emergency situations, the Host NSA may take appropriate emergency measures to prevent irregularities committed on its territory. In this case, it shall without delay inform the Home NSA. These measures shall be notified in writing simultaneously to the head office of the undertaking and, where appropriate, to the branch concerned. The measures may, in particular, include a prohibition on the undertaking from continuing to conclude new contracts in the territory of the Member State concerned, or any other measure provided for by national legislation. The rationale for the invocation of the measures must be explained in the notification. The notification may be drafted in the language of the Host State.

4.3 College of Supervisors and collaboration platforms

4.3.1 Where a college of supervisors is established

- 4.3.1.1 Upon its own initiative or following a reasoned request from the Supervisory Authority responsible for the supervision of a branch, where the branch meets at least one of the conditions in points (a) or (b) of Article 354(1) of Commission Delegated Regulation (EU) 2015/35²⁴ ("Delegated Regulation"), the group supervisor shall invite that supervisory authority to participate in any relevant activity of the college of supervisors.
- 4.3.1.2 The group supervisor shall consider, where relevant, inviting Host NSAs where the undertaking pursues business under FoS to specific sessions where activity carried out by way of FoS are being discussed.
- 4.3.1.3 Requests to participate in the meetings of a college of supervisors may come from the group supervisor, a member or participant of the college or by the Host NSA where the undertaking pursues business under FoS. These requests shall be assessed by the group supervisor.

²⁴ COMMISSION DELEGATED REGULATION (EU) 2015/35 of 10 October 2014 supplementing Directive2009/138/EC of the European Parliament and of the Council on the taking-up and pursuit of the business of Insurance and Reinsurance (Solvency II) (OJ L 12, 17.01.2015. p.1)

4.3.2 Collaboration platforms

- 4.3.2.1 In the absence of a college of supervisors the Home and Host NSAs shall consider agreeing on setting up collaboration platform in accordance with Article 152b(2) of Solvency II Directive in order to enhance efficiency of monitoring the activity carried out in the Host Member State through a branch or by way of FoS. Home NSA shall inform EIOPA on setting up of collaboration platform and update EIOPA on the important developments in order to have an overview of ongoing collaboration platforms and to be able to coordinate the work therein in timely manner in case of need.
- 4.3.2.2 EIOPA may, in the case of justified concerns about negative effects on policy holders, on its own initiative or at the request of one or more of the relevant supervisory authorities, set up and coordinate a collaboration platform to strengthen the exchange of information and to enhance collaboration between the relevant supervisory authorities where an insurance or reinsurance undertaking carries out, or intends to carry out, activities through a branch or by way of FoS and where:
 - a) Such activities are of relevance with respect to the host Member State's market. The significance of the cross-border insurance activity should be assessed on a case-by-case basis:
 - a. in terms of the annual gross written premium subscribed in the host Member State compared to the total annual gross written premiums of the undertaking,
 - b. in terms of the impact on the policyholder protection in the host Member State and
 - c. in terms of the impact of the branch or activity of the respective undertaking on the market of the host Member State.
 - b) A notification by the Home NSA of deteriorating financial conditions or other emerging risks which may have a cross-border effect has been made to EIOPA under Article 152a(2) of Solvency II Directive in accordance with Section 4.2.1; or
 - c) The matter has been referred to EIOPA under Article 152a(2) of Solvency II Directive in case where the Host NSA(s) notifies the relevant Home NSA of serious and reasoned concerns with regard to consumer protection in accordance with Section 4.2.3 and where no bilateral solution can be found.

- 4.3.2.3 In cases specified in the previous paragraph, the allocation of tasks and responsibilities between the Home NSA and Host NSA shall not prevent respectively:
 - a) the Host NSA from utilising the knowledge of the Home NSA about the prudential status of the undertaking in order to achieve effective supervision and adequate protection of policyholders;
 - b) the Home NSA from utilising the knowledge of the Host NSA about the local market and risks in order to achieve effective supervision and adequate protection of policyholders.

Cooperation and coordination in a collaboration platform shall include at least the following:

- a) regular meetings and teleconferences with the involvement of the Home NSA all relevant Host NSAs and EIOPA when appropriate;
- b) regular exchange of information as follows, covering both prudential and conduct of business aspects:
 - a. the Home NSA to provide to the Host NSA and EIOPA when appropriate the conclusions drawn by the Supervisory Review Process (SRP) carried out at the level of the undertaking, including information about the financial strength and performance of the undertaking;
 - b. the Home NSA to provide to the Host NSA and EIOPA when appropriate information about the performance of the branch;
 - c. the Host NSAs to provide information on the local markets, their developments and competitive situation as well as on the performance of the industry;
 - d. exchange of any other information necessary for the identification and monitoring of risks originating from the cross-border business model;
- c) meetings of the Home NSA the Host NSA and EIOPA when appropriate with representatives of the undertaking, when necessary;
- d) sharing and delegation of tasks between Home and Host NSAs;
- e) taking any coordinated supervisory actions in a timely manner, where appropriate;

- f) performing joint on-site inspections covering both prudential and conduct of business aspects, where appropriate.
- 4.3.2.4 At the request of EIOPA, the relevant supervisory authorities shall provide all necessary information in a timely manner to allow for the proper functioning of the collaboration platform.
- 4.3.2.5 EIOPA shall coordinate effective and efficient supervisory activities of the collaboration platforms set up by EIOPA and as such, shall be able to initiate and participate in the activities of the collaboration platforms, including on-site inspections, carried out jointly by two or more Supervisory Authorities. However, supervision and enforcement decisions are and remain within the competence of the Home NSA.

4.4 Portfolio transfer

4.4.1 Transfer of portfolio of contracts of insurance undertakings

Article 39 of the Solvency II Directive

- 4.4.1.1 Before an insurance undertaking is authorised under the conditions laid down by its national law to transfer all or part of its portfolio of contracts concluded under the FoE or FoS to an accepting insurance undertaking established within the EEA, the Home NSA of the transferring insurance undertaking shall
 - a) consult the Host NSA of the branch whose portfolio is to be transferred; and
 - b) obtain the agreement of the Supervisory Authority(ies) of the Member State(s) where the contracts were concluded, including the agreement of the Host NSA of the Member State of the branch, in case of risks or commitments in that Member State.
- 4.4.1.2 These opinions and consents shall be given as soon as possible and in any event no later than three months after the date of receipt of the request. The Host NSA shall acknowledge receipt if so requested²⁵. Once this period has expired, if no response was received, the opinion shall be considered positive or consent shall be deemed to have been given.
- 4.4.1.3 When the head office of the accepting insurance undertaking and that of the transferring insurance undertaking are not in the same Member State the Home NSA of the transferring insurance undertaking, shall also obtain a certificate of solvency from the Home NSA of the accepting insurance undertaking, stating

²⁵ Evidence of receipt may assume the form of communications between Supervisory Authorities which take place during the usual course of the portfolio transfer process.

that the accepting insurance undertaking covers the SCR as calculated in accordance with Article 100 of the Solvency II Directive, taking account of the transfer. This certificate shall be issued as soon as possible and in any event no later than three months after receipt of the request.

- 4.4.1.4 In order to facilitate these measures as a whole, the Home NSA of the transferring insurance undertaking shall provide the following minimum information:
 - a) to the Home NSA of the accepting insurance undertaking:
 - a. the draft transfer agreement or the transfer agreement and, if they do not appear in it, the names and addresses of the transferring insurance undertaking and the accepting insurance undertaking, the insurance classes and the details of the nature of the risks or commitments to be transferred;
 - b. the volume of gross and net technical provisions, established on the basis of the contracts to be transferred;
 - c. the volume of gross and net written premiums;
 - d. the volume of the gross and net burden of claims in non-life insurance;
 - e. details of assets transferred;
 - f. details of guarantees provided by the transferring insurance undertaking or a third party (for example a reinsurance undertaking) to safeguard against deterioration of the reserves corresponding to the transferred business; and
 - q. name(s) of the country or countries of the risks or commitments.
 - b) to the Host NSA of the branch whose portfolio is to be transferred:
 - a. the draft transfer agreement or the transfer agreement and, if they do not appear in it, the names and addresses of the transferring insurance undertaking and the accepting insurance undertaking, and the scope of the operation (total or partial transfer of the branch's portfolio);
 - b. arrangements for the settlement of claims in the event of the closure of the branch following the transfer.
 - c) to the Supervisory Authority of the Member State of the risks or commitments:

- a. the draft transfer agreement or the transfer agreement and, if they do not appear in it, the names and addresses of the transferring insurance undertaking and the accepting insurance undertaking.
- 4.4.1.5 If the Home NSA of the accepting insurance undertaking has serious concerns about how that insurance undertaking will perform in the future, it shall inform the Home NSA of the transferring insurance undertaking of those concerns as soon as they arise, but in any event no later than within a period of three months after it has been consulted.
- 4.4.1.6 Where the accepting insurance undertaking:
 - a) has not previously taken up the business of direct insurance and therefore requires authorisation from the Home NSA or requires an extension of its authorisation; and/or
 - b) will cover the risks or commitments through a branch which has yet to be established, or will require an extension of the business which it is entitled to carry on in the State of the Branch; and/or
 - c) will cover the risks or commitments through the provision of services where it has not previously done so;

the relevant Supervisory Authorities shall cooperate to ensure that, as far as possible, their respective functions can be carried out concurrently, to enable the transfer to take place within a reasonable period. As part of this cooperation, the Home NSA of the accepting insurance undertaking shall ensure that the undertaking performs, as deemed necessary, the notifications to establish a branch or to start FoS. The Home NSA of the accepting insurance undertaking shall ensure that the latter notification is performed regardless of the accepting undertaking's intention to write new business in the Host member state(s).

- 4.4.1.7 The Home NSA of the transferring insurance undertaking shall communicate its decision to the Supervisory Authority of the country or countries where the contracts were concluded and the other authorities which were consulted.
- 4.4.1.8 The Supervisory Authorities of the Member States of risks or commitments shall assist the Home NSA of the transferring insurance undertaking or of the accepting insurance undertaking, or the transferring insurance undertaking or the accepting insurance undertaking themselves, at the time of publication of the transfer, in accordance with the law applicable in the Member States of risks or commitments. The Supervisory Authorities shall inform each other about the method of publication of the transfer of portfolio provided by their national law.
- 4.4.1.9 In case of a merger of insurance or reinsurance undertakings, the Supervisory Authorities shall consult each other in accordance with the procedure laid down

for portfolio transfers, and inform each other about the legal consequences of the merger, in particular the validity of existing notifications of cross-border business.

- 4.4.1.10 The authorisation of a transfer under this Section shall not affect the right of Member States to give policyholders the option of cancelling contracts within a fixed period. The Supervisory Authorities shall inform each other of the circumstances and the period within which contracts may be cancelled according to the provisions of their national law.
- 4.4.1.11 Paragraphs 4.4.1.1 to 4.4.1.10 shall apply to the reinsurance business of insurance undertakings.
- 4.4.1.12 The exchange of information between the Supervisory Authorities described in Section 4.4.1 shall be performed by the Home and the Host NSAs by using the IT tool and the templates provided by EIOPA. EIOPA shall be included, for information, in this communication exchange.

4.4.2 Transfer of portfolio of contracts of reinsurance undertakings

Articles 39 of the Solvency II Directive

- 4.4.2.1 Before a portfolio transfer is authorised, when the head office of the transferring reinsurance undertaking and that of the accepting undertaking are not in the same Member State, the Home NSA of the transferring reinsurance undertaking shall obtain a certificate of solvency from the Home NSA of the accepting undertaking, stating that the accepting insurance or reinsurance undertaking covers the SCR, taking the transfer into account, as calculated in accordance with Article 100 of the Solvency II Directive. The certificate of solvency shall, wherever possible, be provided within three months from the date of receipt of the request.
- 4.4.2.2 The Supervisory Authorities concerned by the transfer of the portfolio of contracts shall cooperate to ensure that, as far as possible, their respective functions can be carried out concurrently to enable the transfer to take place in a reasonable period.
- 4.4.2.3 In order to facilitate these measures as a whole, the Home NSA of the transferring reinsurance undertaking shall, wherever possible, provide the following minimum information to the Home NSA of the accepting undertaking within three months from the date of receipt of the request:
 - a) the draft transfer agreement or the final transfer agreement and, if they do not appear in it, the names and addresses of the transferring reinsurance undertaking and the accepting undertaking, and the type of

reinsurance activity provided for under Article 15 (5) of the Solvency II Directive:

- b) the volume of gross and net technical provisions, established on the basis of the contracts to be transferred;
- c) the volume of gross and net written premiums;
- d) the volume of the gross and net burden of claims in non-life reinsurance;
- e) details of assets transferred;
- f) details of guarantees provided by the transferring reinsurance undertaking or a third party to safeguard against deterioration of the reserves corresponding to the transferred business, if relevant.
- 4.4.2.4 If the Home NSA of the accepting undertaking has serious concerns about how that accepting undertaking will perform in the future, it shall inform the Home NSA of the transferring reinsurance undertaking of those concerns as soon as they arise, but in any event no later than within a period of three months after it has been consulted.
- 4.4.2.5 Where the accepting undertaking:
 - a) has not previously taken up the business of reinsurance and therefore requires authorisation from the Home NSA or requires an extension of its authorisation; and/or
 - b) will cover the risks or commitments through a branch which has yet to be established, or will require an extension of the business which it is entitled to carry on in the Member State of the branch;

the relevant Supervisory Authorities shall co-operate to ensure that, as far as possible, their respective functions can be carried out concurrently, to enable the transfer to take place within a reasonable period.

- 4.4.2.6 In the case of a merger of reinsurance undertakings, the Supervisory Authorities shall consult each other in accordance with the procedure laid down for portfolio transfers.
- 4.4.2.7 The exchange of information between the Supervisory Authorities described in Section 4.4.2 shall be performed by the Home and the Host NSAs by using the IT tool and the templates provided by EIOPA. EIOPA shall be included, for information, in this communication exchange.

4.4.3 Transfer of portfolio of contracts of insurance branches of Third Countries

Article 164 of the Solvency II Directive

- 4.4.3.1 Under the conditions provided for by national law, a Supervisory Authority shall authorise the transfer of the portfolio of contracts from a branch of a third Country established in its territory ('Transferor') to an accepting Office established in a Member State of the EEA ('Accepting Office'), subject to the following conditions:
 - a) the Supervisory Authorities of the Home State of the Accepting Office attest that the Accepting Office possesses the necessary solvency;
 - b) if the Accepting Office is a branch of a third country established in the Transferor's Member State, the Supervisory Authority responsible for supervising its overall solvency, shall certify that the Accepting Office possesses the necessary solvency, taking account of the transfer;
 - c) if the Accepting Office is a branch of a third country established in a Member State other than that of the Transferor and where this is authorised in the country of establishment of the Transferor, the Supervisory Authorities of the country of establishment of the Accepting Office or those in charge of supervising its solvency, must certify that the Accepting Office possesses the necessary solvency, taking account of the transfer;
 - d) the Supervisory Authorities of the Member State of establishment of the Accepting Office must certify that the law of that Member State permits such a transfer, and that the Member State has agreed to the transfer;
 - e) that the Supervisory Authorities of the Member State of the risks and those of the Member State of the commitment, shall give their agreement within a period of three months from the request for an opinion from the Competent Authority of the Home State of the transferring Undertaking. If the Supervisory Authorities have not responded at the expiry of the three-month period, agreement will be assumed.

4.5 Supervisory measures

4.5.1 Restriction or prohibition of free disposal of assets

Article 137, 138, 139, 140 of the Solvency II Directive

- 4.5.1.1 The Home NSA shall determine whether to initiate any measure(s) to restrict or prohibit the free disposal of the assets of an insurance or reinsurance undertaking. The Home NSA shall inform the Host NSA and EIOPA of its intention to initiate any such measure(s). The notification and the measure may happen simultaneously where immediate action is required.
- 4.5.1.2 The Home NSA shall, in writing, inform the Supervisory Authorities of the relevant Member States where the assets are located, whose disposal must be restricted or prohibited.
- 4.5.1.3 Insofar as it lies within their possibilities, The Supervisory Authorities of the Member States concerned shall check the existence or the location of the assets previously identified by the Home NSA, on the basis of information provided by the latter, and on its request.
- 4.5.1.4 The Supervisory Authority to which the request has been made shall restrict or prohibit the disposal of the assets and inform the Home NSA of the measures taken. It shall endeavour to provide any assistance required by the Home NSA.
- 4.5.1.5 The Home NSA may perform the notification described at Section 4.5.1 through the IT tool provided by EIOPA and where applicable as part of the process of notification described in Section 4.2.1

4.5.2 Recovery plan

Article 138, 141, 142 of the Solvency II Directive

- 4.5.2.1 When an undertaking submits a recovery plan to the Home NSA, the Home NSA shall assume responsibility for approving and ensuring the implementation of the recovery plan.
- 4.5.2.2 The Home NSA shall inform the Host NSA and EIOPA that the undertaking is required to submit a recovery plan. It shall inform the Host NSA about the measures referred in Article 138(3) of the Solvency II Directive to achieve, within six months from the observation of non-compliance with the SCR, the reestablishment of the level of eligible own funds covering the SCR or the reduction of its risk profile to ensure compliance with the SCR, together with any information which it considers appropriate in the circumstances, on the request of the Host NSA.

- 4.5.2.3 If Supervisory Authorities from more than one Member State consider that an exceptional adverse situation, as referred to in the second sub-paragraph of Article 138 (4), of the Solvency II Directive, exists in their markets, that affects or may affect undertakings in their markets, those Supervisory Authorities shall share information and shall cooperate, where appropriate, to submit²⁶, a joint request to EIOPA to declare such an exceptional situation to exist in their respective markets.
- 4.5.2.4 The Home NSA may perform the notification described at Section 4.5.2 through the IT tool provided by EIOPA and where applicable as part of the process of notification described in Section 4.2.1

4.5.3 Short-term finance scheme

Article 139, 141, 142 of the Solvency II Directive

- 4.5.3.1 When an insurance or reinsurance undertaking is required to provide the Home NSA with the short-term realistic financial scheme, the Home NSA shall assume responsibility for approving and ensuring the implementation of the plan.
- 4.5.3.2 The Home NSA shall inform the Host NSA and EIOPA that the insurance or reinsurance undertaking submitted a short-term finance scheme. It shall inform the Host NSA about the measures to restore, within three months of that observation, the eligible basic own funds, at least to the level of the MCR or to reduce its risk profile to ensure compliance with the MCR, together with any information which it considers appropriate in the circumstances, on the request of the Host NSA.
- 4.5.3.3 The Home NSA may perform the notification described at Section 4.5.3 through the IT tool provided by EIOPA and where applicable as part of the process of notification described in Section 4.2.1

4.5.4 Withdrawal or lapse of authorisation

Article 144 of the Solvency II Directive

- 4.5.4.1 The Home NSA shall assume sole responsibility for terminating the activity of an undertaking. This shall be without prejudice to the competences of Host NSA.
- 4.5.4.2 Except under special circumstances and before withdrawing authorisation in respect of one, several or all the classes for which the undertaking is authorised,

²⁶ In the meaning of paragraph 24 of Processes for declaring exceptional adverse situation for the purposes of Article 138 of Solvency II (Extension of the Recovery Period) (EIOPA-BoS-16/158-rev1 8 August 2016).

the Home NSA shall inform the Supervisory Authorities of all affected Host States of its intention.

- 4.5.4.3 The Home NSA shall communicate the decision on withdrawal or the lapse of authorisation to the Supervisory Authorities of all other Member States in respect of which the undertaking has exercised rights under FoE or FoS and EIOPA, and update its website accordingly.
- 4.5.4.4 The Supervisory Authorities shall, throughout the process relating to the withdrawal or lapse of authorisation, collaborate in good faith and in mutual recognition of the competence of each Supervisory Authority.
- 4.5.4.5 The Supervisory Authorities shall act collectively and collaboratively, to ensure a coordinated response to the consequences of the withdrawal or the lapse of authorisation.
- 4.5.4.6 The Home NSA may perform the notification described at Section 4.5.4 through the IT tool provided by EIOPA and where possible as part of the process of notification described in Section 4.2.1

4.5.5 Application of reorganisation measures and winding-up proceedings of an insurance undertaking

Articles 269, 270, 273, 296 of the Solvency II Directive

- 4.5.5.1 A decision adopted according to the Home Member State's legislation concerning reorganisation measures or the opening of winding-up proceedings of an insurance or reinsurance undertaking, including its branches in other Member States, shall be recognised without any further formalities within the territory of all other Member States and shall be effective there as soon as the decision is effective in the Member State in which the proceedings are opened.
- 4.5.5.2 The Supervisory Authority of the Home State shall assume sole responsibility for reorganising or terminating the activity of an (re)insurance undertaking.
- 4.5.5.3 The Home NSA shall communicate, as a matter of urgency, where possible before the adoption of such a measure and failing that immediately thereafter, the decision on the winding-up or adoption of reorganisation measures to all other Supervisory Authorities and EIOPA. The information shall include the possible practical effect of such measures.
- 4.5.5.4 When an insurance or reinsurance undertaking, whose head office is situated outside EEA, has branches established in more than one Member State, each branch shall be treated independently with regard to the application of Title IV of the Solvency II Directive (Reorganisation and winding-up of insurance undertakings).
- 4.5.5.5 The Supervisory Authorities of the relevant Member States shall endeavour to coordinate their actions.

5 PART V REGULAR EXCHANGE OF QUANTITATIVE DATA

5. Exchange of quantitative information between the Home NSA and Host NSA

Article 159 of the Solvency II Directive

- 5.1 The following information, extracted from the Solvency II quantitative reporting templates ("QRTs"), shall be derived from EIOPA Central Repository and made available by undertaking to each concerned Host NSA through EIOPA HUB²⁷ following the deadlines established for this tool:
 - a) premiums written by line of business ("LoB"), split by:
 - a. business underwritten in the Host country through a branch established in the Host country;
 - b. business underwritten through FoS by the branch established in the Host Country;
 - c. business underwritten in the Host country through FoS by the undertaking or any EEA branch.
 - b) claims incurred and commissions by LoB, split by a) i. and ii.
 - c) frequency of claims for motor vehicle liability (except carrier's liability), split by a) i. and ii;
 - d) average cost of claims for motor vehicle liability (except carrier's liability), split by a) i. and ii.
 - e) This information shall be extracted from the appropriate QRTs reported by undertakings.

In addition to the above, to keep Host NSAs informed of which notifications are active and which are inactive, EIOPA shall include in the report developed in accordance with the first sub-paragraph, a table for each individual country all entries from the register on "FoS", "Branch" and "FoS by Branches", where the country is the host country. In addition to the pure extraction from the EIOPA register EIOPA to add a flag on "active/inactive" based on SII reporting data.

²⁷ EIOPA HUB is a technical file transfer system established to collect submissions from competent authorities.

PART V REGULAR EXCHANGE OF QUANTITATIVE DATA

Furthermore, EIOPA shall make available to the NSA through EIOPA HUB following the deadlines established for this tool a table with aggregated data on gross written premiums grouped at the level of LoB per each Member State.

- 5.2 Host NSAs may ask the Home NSA whether the information for the Host NSA is included in other cross-border QRTs templates reported by the undertaking to the Home NSA, and ask to be made available through EIOPA HUB specific information from those templates together with the information identified in paragraph 5.1.
- 5.3 In order to improve the quality of the information exchanged in accordance with paragraph 5.1 and on a 'need to know' basis, EIOPA may add to the standard report described in 5.1 additional information or more granular data where appropriate.
- 5.4 The above paragraphs do not prevent Supervisory Authorities from exchanging more granular data on a more regular basis or upon request

6 PART VI HANDLING OF POLICYHOLDER COMPLAINTS

6.1 Complaints handling systems in Member States

Articles 183 and 185 (3) (I) of the Solvency II Directive

- 6.1.1 In the context of the development of the single market in insurance, it is important that all complaints received by Supervisory Authorities from policyholders²⁸ are dealt with by the relevant complaints handling body, irrespective of whether the complaint involves policyholders and insurers in different Member States.
- 6.1.2 The insurance undertaking shall inform the policy holder of the arrangements for handling complaints of policy holders concerning contracts including, where appropriate, the existence of a complaints body, without prejudice to the right of the policy holder to take legal proceedings.
- 6.1.3 The manner in which complaints handling received by Supervisory Authorities²⁹ is organised within a jurisdiction is a matter for each jurisdiction and therefore not subject to this Decision. There are a wide variety of complaints handling systems in Member States. In some Member States, complaints are dealt with by Supervisory Authorities, whilst others have developed Ombudsman services or other systems, or have a combination of both. The ways in which these systems operate vary considerably across the Member States.
- 6.1.4 Where complaints have a cross-border component, it is important for individual policyholders and for consumer confidence generally, that the differences in complaints handling systems do not prevent or deter complaints from being directed to the relevant bodies and subsequently dealt with.
- 6.1.5 The procedures outlined in this Decision are intended to facilitate the process of identifying the relevant complaints handling body and ensure that complaints filed with a Supervisory Authority are addressed and arrive at the body that is competent to deal with the complaint.

²⁸ The term 'policyholder' shall be understood in a broad sense, including all persons that file an insurance-related complaint.

²⁹ This Decision focuses on complaints received by Supervisory Authorities, and does not concern consumer complaints received by the undertakings or intermediaries, which are covered by the relevant EIOPA guidelines.

6.1.6 It is recognised that the Commission has set up a network (FIN-NET)³⁰ to facilitate the out-of-court resolution of consumer complaints where the service provider is established in a Member State other than the Member State where the consumer is domiciled. Nothing in this Decision is intended to interfere with this agreement or the process developed under FIN-NET.

6.2 Complaints handling procedures

- 6.2.1 A complaint shall be assessed by the Supervisory Authority which initially receives it, to establish which body is competent to deal with the complaint. Where possible, the Supervisory Authority shall keep the complainant informed about further handling of the complaint.
- 6.2.2 In cases where the Supervisory Authority that receives the complaint is competent to deal with the complaint, it shall inform the complainant, as soon as possible, of applicable national procedures.
- 6.2.3 In cases where the Supervisory Authority that receives the complaint is not competent to deal with the complaint, but another body in the same jurisdiction is, the Supervisory Authority shall, as soon as possible:
 - if legally possible, pass the complaint directly to the body competent to deal with the complaint and inform the complainant of the referral;
 - if legally not possible, provide the complainant with any relevant information about the body competent to deal with the complaint, to facilitate the referral of the complaint by the complainant.
- 6.2.4 In cases where the Supervisory Authority that receives the complaint is not competent to deal with the complaint, but a Supervisory Authority or a competent body in another jurisdiction is, the Supervisory Authority that receives the complaint shall, as soon as possible:
 - if legally possible, pass the complaint directly to the Supervisory Authority or competent body in the other jurisdiction and inform the complainant of the referral;
 - if legally not possible, provide the complainant with any relevant information about the Supervisory Authority or competent body in the other jurisdiction, in order to facilitate the referral of the complaint by the complainant;

-

³⁰ http://ec.europa.eu/fin-net

PART VI HANDLINGS OF POLICYHOLDER COMPLAINTS

- in case that the complaint was passed to a competent body other than the Supervisory Authority, inform, if legally possible, the Supervisory Authority of the other jurisdiction of the complaint.
- 6.2.5 If a complaint contains issues relevant to the supervision of an undertaking, which operates under the freedom of establishment or FoS, the Supervisory Authority that deals with the complaint shall, as soon as possible, and if legally possible, provide the Supervisory Authority of the Home State with relevant information on the complaint.
- 6.2.6 In the cases referred to in paragraphs 6.2.3 to 6.2.4, the Supervisory Authority that initially receives the complaint shall explain to the complainant why it considers itself not to be competent to deal with the complaint, and why it considers the other Supervisory Authority or body to be competent.

6.3 Dispute over responsibilities

- 6.3.1 A complaint shall not be passed to another Supervisory Authority or competent body, if the Supervisory Authority or competent body has already declined responsibility for dealing with the complaint.
- 6.3.2 Where differences of opinion exist as to who assumes responsibility for the complaint, every best endeavour shall be made to resolve the issue in good faith within four weeks from the date on which the complaint was initially received from the complainant.
- 6.3.3 If the question of responsibility cannot be settled, the Supervisory Authority which initially received the complaint from the complainant may refer the matter to EIOPA for the purpose of settling the disagreement through non-binding mediation.

6.4 Information to the public

- 6.4.1 The Supervisory Authorities shall provide the following information³¹ about their national complaints handling system, including updates, as necessary:
 - a) Authorities/bodies competent to deal with policyholder complaints (indicating, in particular, whether it is an ombudsman or a supervisory service);

³¹ This paragraph does not impose a requirement on the Supervisory Authority to provide the information in a particular format.

PART VI HANDLINGS OF POLICYHOLDER COMPLAINTS

- b) Contact information for the Supervisory Authority/body competent to deal with policyholder complaints (name, postal address, phone number, fax number, any e-mail address, any website address);
- c) Organisation for the Supervisory Authority/body competent to deal with policyholder complaints (e.g. whether it is a statutory/voluntary system, any charges payable by the complainant);
- d) Coverage for the Supervisory Authority/body competent to deal with policyholder complaints (e.g. insurers/intermediaries covered, insurance products covered);
- e) Competence for the Supervisory Authority/body competent to deal with policyholder complaints (e.g. out of court settlement/system that does not issue decisions; information on whether the decision is binding or not);
- f) Preconditions of the Supervisory Authority/body competent to deal with policyholder complaints (e.g. necessity to address the undertaking in the first place);
- g) Restrictions of the Supervisory Authority/body competent to deal with policyholder complaints (e.g. any limit on the amount awarded, time limits in bringing the complaint to the Supervisory Authority, restrictions on the type of complainant - professional/non-professional, consumer associations/other third parties);
- 6.4.2 The Supervisory Authorities shall make the aforementioned information available through their public website.³² The Public Area of EIOPA's websiteshall provide links to the relevant national websites.

67/101

³² A Supervisory Authority may make the information in paragraph 6.4.1 available by way of a link on their website to the relevant information.

7 PART VII OTHER ISSUES

7.1 Assistance (class 18)

Article 197 of the Solvency II Directive

- 7.1.1 When an undertaking is authorised in class 18 in Part A of Annex I of the Solvency II Directive the Host NSA shall collaborate with the Home NSA, to verify the resources available to the undertaking to successfully carry out assistance operations, in so far as national legislation provides for supervision of these resources.
- 7.1.2 In order to carry out this supervision, the Supervisory Authorities of the Home State and the Host State shall exchange, on request, any information relating to assistance resources available to the undertaking, and communicate to each other the necessary documents and information for the exercise of the supervision.

Done at Frankfurt am Main on 10.06.2021.

ANNEXES

ANNEX 1: Certificate of Solvency

CERTIFICATE OF SOLVENCY

ISSUED BY THE SUPERVISORY AUTHORITY OF THE HOME STATE

relating to(registered name of the Undertaking, legal form ³³ , head office)
issued by(Supervisory Authority)
PART I
The issuing Supervisory Authority certifies that the above mentioned undertaking is entitled to carry on the following classes or groups of classes:
A. Life assurance
Classes (according to Article 15(2) and as set out in Annex II to the Solvency II Directive):
(N°) (Name) since (*) authorisation limited to (**)
B. Non-life insurance
1. Classes (according to Article 15(2) and (3) and as set out in point A of Annex I to the Solvency II Directive):
(No) (Name) since (*) authorisation limited to (**) (subdivision)
(*) Indicate the date of authorisation or the date on which the undertaking began carrying on the class if this preceded compulsory authorisation. When dates are five years and more previously, mention only this fact. (**) Indicate, where appropriate, the part of the risk to which authorisation is limited, as provided for in the second paragraph of Article 15(3) of the Solvency II Directive.

³³ As per Annex III to the Solvency II Directive

ANNEXES

2. Groups of classes (according to Article 15(2) and (3) and as set out in point B of Annex I to the Solvency II Directive):
(letter) (Name) since (*) authorisation limited to (**)
C. Reinsurance
Type of authorisation granted in accordance with Article 15(5) of the Solvency II Directive:
(Name) since (*) authorisation limited to (**)
PART II
The issuing Supervisory Authority certifies that the above mentioned undertaking covers the Solvency Capital Requirement and Minimum Capital Requirement calculated in accordance with Articles 100 and 129 of the Solvency II Directive.
The undertaking possesses the necessary Solvency Capital Requirement and Minimum Capital Requirement, taking into account the portfolio transfer. (***)
The above attestations and information are valid at the date of issuance of this certificate of solvency.
(date, signature, position)
(*) Indicate the date of authorisation or the date on which the Undertaking began carrying on the class if this preceded compulsory authorisation. When dates are five years and more previously, mention only this fact. (**) Indicate, where appropriate, the part of the risk to which authorisation is limited, as provided for in the second paragraph of Article 15(3) of the Solvency II Directive.

^(***) Delete as appropriate

ANNEX 2: Table summarising the information flows between the Supervisory Authorities foreseen by this Decision

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Authorisation	A.1	2.2	Exchange of authorisation information with other insurance supervisory authorities	Home NSA	Host NSA(s)	26(1)	No	na	Prior to granting an authorisation	2 Weeks
Authorisation	A.2	2.3	Exchange of authorisation information with other financial supervisory authorities	Home NSA	Other Financial Supervisory Authorities	26(2)	No	na	Prior to granting an authorisation	na
Authorisation	A.3	2.4	Exchange of info on all persons who effectively run the undertaking, key function, shareholders, member with qualified holdings	Home NSA	Host NSA(s)	26(3)	No	na	Prior to granting an authorisation	2 Weeks
Authorisation	A.4	2.5	Exchange of info on authorisation sought in other Member States from the Home Supervisory Authority with Insurance Supervisory Authorities of other Member States	Home NSA	Host NSA(s)	na	No	na	Prior to granting an authorisation	2 Weeks
Authorisation	A.5	2.6.4.1	Notification on cross border activity	Home NSA	Host NSA(s) EIOPA	152a(1)	Yes	Intention to authorise	Prior to granting an authorisation	A.6 & A.7
Authorisation	A.6	2.6.5.1	Answer to the notification of the intention to authorise a cross border undertaking	Host NSA	Home NSA EIOPA	na	Yes	Answer from Host NSA	1 Month after A.5	
Authorisation	A.7	2.6.5.3	EIOPA's feedback on the notification of the intention to authorise a cross border undertaking	EIOPA	Home NSA Host NSA(s)	na	Yes	EIOPA's view	1 Month after A.5	
Authorisation	A.8	2.6.5.5	End of authorisation process	Home NSA	Host NSA(s) EIOPA	na	Yes	End of authorisation process	1 Month after the end of the authorisation process	

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency 11 article		Template if in scope	Timing to send	Timing to reply
On-going (event specific)	A.9	2.7	Exchange of info on Acquisitions and increase of shareholdings in undertakings	Home NSA	Host NSA(s)	60	No		na	2 Weeks
Insurance Branch	IB.1	NA	Notification of the intention to establish a branch in another Member State by an insurance undertaking	Insurance Undertaking	Home NSA	145(1)	No		na	See § 3.1.4
Insurance Branch	IB.2	3.1.1.1 3.1.1.2 and 3.1.1.3	Communication of notification to establish a branch	Home NSA	Host NSA(s) EIOPA	146(1)	Yes	Communication of notification Insurance Branch	3 Months after receiving IB.1	IB.3 and IB.5
Insurance Branch	IB.3	3.1.1.4	Ad hoc request on undertaking which intends to establish a branch	Host NSA	Home NSA EIOPA	na	Yes	Answer from Host NSA (ad hoc requests)	1 Month after receiving IB.2	na
Insurance Branch	IB.4	3.1.1.6	Informal exchange on undertaking operating in FoE (before the Home NSA sends the complete notification)	Home NSA	Host NSA	na	Yes	Informal exchange	Before IB.2	Before IB.2
Insurance Branch	IB.5	3.1.2.1	Information to be provided as "answer" to Communication of notification to establish a branch / changes to branch	Host NSA	Home NSA EIOPA	146(3)	Yes	Answer from Host NSA general answer	2 Months after receiving IB.2	na
Insurance Branch	IB.6	3.1.2.4	EIOPA's view on the communication of notification	EIOPA	Home NSA Host NSA(s)		Yes	EIOPA's view	2 Months after receiving IB.2	na
Insurance Branch	IB.7	3.1.3	Information of Communication of notification to establish a branch to the Host NSA	Home NSA	Insurance Undertaking	na	No	na	3 Months after receiving IB.1	na
Insurance Branch	IB.8	3.1.5.1	Information of change(s) to the insurance branch	Insurance Undertaking	Home NSA	145(4)	No	na	1 month before making the change(s)	ena
Insurance Branch	IB.9	3.1.5.2	Information of change(s) to the insurance branch	Home NSA	Host NSA(s) EIOPA	145(4)	Yes	Changes to branch	1 Month after receiving IB.7	na
Insurance Branch	IB.10	3.1.6.1	Notification in case of discontinuation of branch activities (to be applied whenever a branch no longer accepts new business and the undertaking no longer admin its portfolio of contracts through the branch)	Home NSA	Host NSA(s) EIOPA	na	Yes	Branch discontinuation	Before the discontinuation	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency 11 article		Template if in scope	Timing to send	Timing to reply
Insurance Branch	IB.11	3.1.6.4	Notification to confirm the closure of the branch	Home NSA	Host NSA(s) EIOPA	na	Yes	Branch closure	1 Month after the closure	IB.12
Insurance Branch	IB.12	na	Information to be provided as "answer" to confirmation of the closure of the branch	Host NSA(s) EIOPA	Home NSA	na	Yes	Confirmation of closure/discontinuation	After IB.11	na
Reinsurance Branch	RB.1	na	Notification of the intention to establish a branch in another Member State by an reinsurance undertaking	Reinsurance undertaking	Home NSA	na	No	na	na	See § 3.1.7
Reinsurance Branch	RB.2	3.1.7.2	Communication of notification to establish a reinsurance branch	Home NSA	Host NSA(s) EIOPA	na	Yes	Communication of notification reinsurance Branch	1 Month after RB.1	
Reinsurance Branch	RB.3	NA	(Optional) reply following the Communication of notification to establish a reinsurance branch	Host NSA EIOPA	Home NSA EIOPA	na	Yes	Answer from Host NSA general answer (RE) EIOPA's view		
Reinsurance Branch	RB.4	3.1.7.3	Information of change(s) to the reinsurance branch	Home NSA	Host NSA(s) EIOPA	na	Yes	Changes to branch (RE)	1 Month after receiving RB.4	na
Reinsurance Branch	RB.5	3.1.7.4	Notification in case of discontinuation of reinsurance branch activities (to be applied whenever a branch no longer accepts new business and the undertaking no longer admin its portfolio of contracts through the branch)	Home NSA	Host NSA(s) EIOPA	na	Yes	Branch discontinuation (RE)	Before the discontinuation	na
Reinsurance Branch	RB.6	3.1.7.5	Notification to confirm the closure of the reinsurance branch	Home NSA	Host NSA(s) EIOPA	na	Yes	Branch closure (RE)	1 Month after the closure	RB.7
Reinsurance Branch	RB.7	na	Information to be provided as "answer" to confirmation of the closure of the branch	Host NSA(s) EIOPA	Home NSA	na	Yes	Confirmation of closure/discontinuation	After RB.6	na
Freedom of Services	F.1	3.2.1.1	Notification of the insurance undertaking indicating the nature of the risks or commitments it proposes to cover in another MS in FoS	Insurance Undertaking	Home NSA	147	No	na	na	See § 3.2
Freedom of Services	F.2	3.2.1.1 3.2.1.2 and 3.2.1.3	Communication of notification to start FoS	Home NSA	Host NSA(s) EIOPA	148	Yes	Start FoS	ASAP max 1 Month after F.1	F.3 and F.4

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Freedom of Services	F.3	3.2.1.4	Ad hoc request on undertaking starting FoS following communication of notification to start FoS	Host NSA	Home NSA EIOPA	na	Yes	Answer from Host NSA (ad hoc requests)	ASAP after F.2	na
Freedom of Services	F.4	3.2.1.6 and 3.2.1.7	Informal exchange on undertaking operating in FoS (before the Home NSA sends the complete notification)	Home NSA	Host NSA(s)	na	No	na	Before F.2	na
Freedom of Services	F.5	3.2.2.1	Information to be provided as "answer" to Communication of notification to start FoS	Host NSA	Home NSA EIOPA	na		Answer from Host NSA general answer	ASAP after F.2	na
Freedom of Services	F.6	3.2.2.4	EIOPA's view on the communication of notification	EIOPA	Home NSA Host NSA(s)	na	Yes	EIOPA's view	ASAP after F.2	na
Freedom of Services	F.7	3.2.3.1.2	Information that the communication of notification to start FoS has been sent to the Host NSA	Home NSA	Insurance Undertaking	na	No	na	Immediately after F.2	na
Freedom of Services	F.8	3.2.3.2	Information on any conditions, under which, in the interests of the general good, the activity must be pursued within the territory of the Host State OR communication that no conditions are imposed from Host Supervisory Authority to	Host NSA	Insurance Undertaking	na	No	na	ASAP after F.2	na
Freedom of Services	F.9	3.2.5.1	Information of change(s) to the FoS	Insurance Undertaking	Home NSA	149	No	na		
Freedom of Services	F.10	3.2.5.2	Information of change(s) to the FoS	Home NSA	Host NSA(s) EIOPA	149	Yes	Changes to FoS		
Freedom of Services	F.11	3.2.6.1	Notification if business activities will no longer be provided by FoS in the Host Member State. This shall include stop of writing new business, termination of the existing policies and settlement of the outstanding claims (if information is available)	Home NSA	Host NSA(s) EIOPA	na	Yes		Before the discontinuation	na
Freedom of Services	F.12	3.2.6.2	Notification to confirm the discontinuation of FoS business	Home NSA	Host NSA(s) EIOPA	na	Yes	()	1 Month after the discontinuation	F.13
Freedom of Services	F.13	na	Information to be provided as "answer" to confirmation of the discontinuation of FoS business	Host NSA(s) EIOPA	Home NSA	na	Yes			na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Health as alternative to Social Security	H.1	3.3.2.1	Information of contracts covering the risks under Class 2 (Sickness), where these contracts may serve as a partial or complete alternative to health cover provided by the statutory social security system	Host NSA	Home NSA	206(1)	No		As soon as existing legislation is changed or new legislation is introduced	na
Health as alternative to Social Security	H.2	3.3.2.2	Information of contracts covering the risks under Class 2 (Sickness), where these contracts may serve as a partial or complete alternative to health cover provided by the statutory social security system	Home NSA	Insurance Undertaking	206(1)	No	na	After H.1	na
Health as alternative to Social Security	H.3	3.3.2.3	Link to the websites where the insurance classes and/or insurance products are listed, for which policy conditions have to be presented to the Host NSA	Home NSA	EIOPA	206(1)	No		H.1	na
Notification from Home NSA (UD)	UD.1	4.2.1.9	Notification when an undertaking carrying out cross-border business in the form of FoE and/or FoS shows deteriorating financial conditions / other emerging risks which may have cross-border effects	Home NSA	Host NSA(s) EIOPA	152a(2)		emerging risks	Within 1 month after detection deteriorating financial conditions	
Notification from Home NSA (UD)	UD.2	4.2.1.11	EI OPA Answer to Home NSA notification when undertaking carrying out cross-border business in the form of FoE and/or FoS shows deteriorating financial conditions or other emerging risks which may have cross-border effects	EIOPA	Home NSA optional Host NSA(s)	152a(2)		Reply to Deteriorating financial conditions or other emerging risks	- 1	NT.3 or NT.4
Notification from Home NSA (UD)	UD.3	4.2.1.11	Host NSA Answer to Home NSA notification when undertaking carrying out cross-border business in the form of FoE and/or FoS shows deteriorating financial conditions or other emerging risks which may have cross-border effects	Host NSA	Home NSA optional Other Host NSA(s) EIOPA	152a(2)	Yes	Reply to Deteriorating financial conditions or other emerging risks	after NT.1 or	na
Notification from Home NSA (UD)	UD.4	4.2.1.12	Relevant updates on the notified undertaking	Home NSA	Host NSA(s) EIOPA	na	Yes	Relevant updates	Optional	na
Information from the Home NSA	IHM.1	4.2.2.1 a	Changes to the assessment of the suitability of shareholders and members	Home NSA	Host NSA(s) optional EIOPA	26	No	na	In a timely manner	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Information from the Home NSA	IHM.2		Details on undertakings showing deteriorating financial conditions or other emerging risks not reported due the application of the materiality thresholds described at paragraph 4.2.1	Home NSA	Host NSA(s) optional EIOPA	152a(2)	No	na	manner	na
Information from the Home NSA	IHM.3	4.2.2.3	Home NSA decides to carry out on-site inspections in a branch situated in another member state	Home NSA	Host NSA(s) optional EIOPA	33	No		4 weeks before the on-site inspection date	IHM.4
Information from the Home NSA	IHM.4	4.2.2.5	Changes of scope of on-site inspection in a branch situated in another member state	Home NSA	Host NSA(s) optional EIOPA	na	No	na	na	IHM.4
Information from the Home NSA	IHM.5	4.2.2.6	Reply following the Communication of intention to carry out a on-site inspection in a branch	Host NSA	Host NSA(s) optional EIOPA	33	No	na	After IHM.2 without delay	na
Information from the Home NSA	IHM.6	4.2.2.7	Observations from the investigation, and any consequences that may arise	Home NSA	Host NSA(s) optional EIOPA	na	No		After the on- site inspection	na
Notification from Host NSA (CCP)	CCP.1	4.2.3.6	Notification when an undertaking carrying out cross-border business poses serious and reasoned concerns with regard to consumer protection	Host NSA	Home NSA EIOPA optional: Host NSA(s)	152a(2)		consumer protection	Within 1 month after detection of concerns consumer protection	
Notification from Host NSA (CCP)	CCP.2	4.2.3.8	EI OPA Answer to Host NSA notification when undertaking carrying out cross-border business poses serious and reasoned concerns with regard to consumer protection	EIOPA	Home NSA optional Host NSA(s)	152a(2)		EIOPA's reply to Concerns consumer protection	Optional	na
Notification from Host NSA (CCP)	CCP.3	4.2.3.8	Host NSA Answer to Host NSA notification when undertaking carrying out cross-border business poses serious and reasoned concerns with regard to consumer protection	Host NSA	Home NSA optional Other Host NSA(s) EIOPA	152a(2)		· · · · · · · · · · · · · · · · · · ·	Within 1 month after CPP.1	na
Notification from Host NSA (CCP)	CCP.4	4.2.3.9	Relevant updates on the notified undertaking	Home NSA	Host NSA(s) EIOPA	na	Yes	Relevant updates	Optional	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Information from the Host NSA	IHS.1	4.2.4.1	Situations that can influence the Home NSA's assessment of the suitability of shareholders and members with qualifying holdings	Host NSA	Home NSA optional EIOPA Host NSA(s)	26	No		At the earliest opportunity	na
Information from the Host NSA	IHS.2	4.2.4.1	Other adverse development that can significantly influence the Home NSA's assessment of the undertaking	Host NSA	Home NSA optional EIOPA Host NSA(s)	30	No		At the earliest opportunity	na
Information from the Host NSA	IHS.3	4.2.4.2	Documents to show compliance to the provision of the host member state	Insurance Undertaking	Host NSA	na	No	na	Upon request	na
Information from the Host NSA	IHS.4	4.2.4.5	Information of the decision to carry out inspections at the Branch or of an outsourced service provider from Host NSA	Host NSA	Home NSA	na	No		4 weeks before the on-site inspection date	IHS.5 & IHS.7
Information from the Host NSA	IHS.5	4.2.4.6	Reply following the Communication of intention to carry out a on-site inspection in a branch	Home NSA	Host NSA	na	no	na	Without delay	na
Information from the Host NSA	IHS.6	4.2.4.7	Request for on-site inspection at the head office of the insurance or reinsurance undertaking	Host NSA	Home NSA	na	No			
Information from the Host NSA	IHS.7	4.2.4.9	Observations from the investigation, and any consequences that may arise	Host NSA	Home NSA optional EIOPA Host NSA(s)	na	No		After the on- site inspection	na
Information from the Host NSA	IHS.8	4.2.4.11	Communication of the infringement(s) observed and the measures to be taken	Host NSA	Insurance Undertaking Authorised agent (branch) Home NSA	na	No	na	na	na
Information from the Host NSA	IHS.9	4.2.4.12	Failure to take action from the undertaking	Host NSA	Home NSA optional EIOPA Host NSA(s)	na	No	na	After IHS.6	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Information from the Host NSA	IHS.10	4.2.4.13	Measures taken by the Host NSA to prevent or penalise further irregularities	Host NSA	Home NSA optional EIOPA Host NSA(s)	na	No	na	After IHS.6 and IHS.7	na
Information from the Host NSA	IHS.11	4.2.4.14	Actions in emergency situation	Host NSA	Home NSA optional EIOPA Host NSA(s)	na	No	na	Without delay	na
Platform of collaboration	CP.1		Home NSA shall inform EIOPA on setting up of collaboration platform	Home NSA	EIOPA	152b	Yes	Set up of platform of collaboration	In a timely manner	na
Platform of collaboration	CP.2	4.3.2.1	Home NSA shall update EIOPA on the important developments in order to have an overview of ongoing collaboration platforms and to be able to provide support in timely manner in case of need.	Home NSA	EIOPA	152b	Yes	Update on platform of collaboration	In a timely manner	na
Portfolio Transfer	PT.1	na	Undertaking request to transfer a portfolio of cross-border contracts	Undertaking	Home NSA of the transferring (re)insurance undertaking	na	No	na	na	na
Portfolio Transfer	PT.2		Information provided to the Host NSA of the Branch whose porfolio is to be transferred (consultation)	Home NSA of the transferring (re)insurance undertaking	Host NSA (only if FoE)	39	Yes	Portfolio transfer Info to Host NSA Branch	Before the authoristation to portfolio transfer	PT.4
Portfolio Transfer	PT.3	4.4.1.1 4.4.1.4	Information provided to the Host NSA of the risks or commitments	Home NSA of the transferring (re)insurance undertaking	Host NSA EIOPA	39	Yes	Portfolio transfer Info to Host NSA risks or commitments	Before the authoristation to portfolio transfer	PT.4
Portfolio Transfer	PT.4	4.4.1.2	Opinion on the portfolio transfer by Host NSAs	Host NSAs	Home NSAs EIOPA	39	Yes	Portfolio Transfer opinion reply	No later than 3 months after PT.2 or PT.3	PT.7

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article		Template if in scope	Timing to send	Timing to reply
Portfolio Transfer	PT.5	4.4.1.4	Information provided to the Home NSA of the accepting insurance undertaking	the transferring (re)insurance	Home NSA of the accepting (re)insurance undertaking EIOPA	39	Yes	Portfolio transfer Info to Home NSA accepting (re)insurance undertaking	No later than 3 months after Undertaking request to transfer portfolio	PT.6
Portfolio Transfer	PT.6		Opinion on the portfolio transfer by Home NSA of the accepting insurance undertaking	Home NSA of the accepting (re)insurance undertaking		39	Yes	Portfolio Transfer opinion reply	No later than 3 months after PT.5	PT.7
Portfolio Transfer	PT.7	4.4.1.7	Communication of the decision on portfolio transfer	Home NSA of the transferring (re)insurance undertaking	the accepting (re)insurance	39	Yes	Communication of the decision on portfolio transfer	After PT.4 and PT.6	PT.8
Portfolio Transfer	PT.8	4.4.1.8	Communication on the the method of publication of portfolio transfer	the accepting (re)insurance undertaking Home NSA of the transferring	transferring (re)insurance undertaking		Yes	Communication of the method of publication of portfolio transfer	After PT.7	
Merger of undertakings	PT.9	4.4.1.9	Consultation in case of merger of (re)insurance undertakings the NSAs shall consult each other in accordance with the procedure laid down for portfolio transfers PT.1 to PT.8	Home NSA Host NSA	Host NSA Home NSA	na	Yes	na	na	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency 11 article		Template if in scope	Timing to send	Timing to reply
Portfolio Transfer third country branch	PTTC.1	4.4.3	Requests to the NSA of the accepting office	Home NSA Host NSA	Host NSA Home NSA	na	Yes	Requests to the NSA of the accepting office	na	PTTC.3
Portfolio Transfer third country branch	PTTC.2	4.4.3	Requests to the NSA of the Member State of the risks or commitments	Home NSA Host NSA	Host NSA Home NSA	164	Yes	Requests to the NSA of the Member State of the risks or commitments	na	PTTC.4
Portfolio Transfer third country branch	PTTC.3	4.4.3	Reply of the NSA of the accepting office	Home NSA Host NSA	Host NSA Home NSA	164	Yes		As soon as available	
Portfolio Transfer third country branch	PTTC.4	4.4.3	Reply of the NSA of the Member State of the risks or commitments	Home NSA Host NSA	Host NSA Home NSA	164	Yes	the Member State of	3 months after receiving PTTC.2	
Portfolio Transfer third country branch	PTTC.5	na	Communication of the decision on portfolio transfer	Home NSA Host NSA	Host NSA Home NSA	164	Yes	Communication of the decision on portfolio transfer	After the decision	
Portfolio Transfer third country branch	PTTC.6	na	Communication on the the method of publication of portfolio transfer	Home NSA Host NSA	Host NSA Home NSA	164	Yes	Communication of the method of publication of portfolio transfer		
Supervisory measures	SM.1	4.4.1	Restriction or prohibition of free disposal of assets	Home NSA	Host NSA(s) EIOPA	137 138 139 140	Yes	Addendum Prohibition free disposal of assets		na
Supervisory measures	SM.2	4.4.2	Recovery Plan	Home NSA	Host NSA(s) EIOPA	138 141 142	Yes	Addendum Recovery Plan	Together with UD.1 or UD.4	na
Supervisory measures	SM.3	4.4.3	Short-term financial scheme	Home NSA	Host NSA(s) EIOPA	139 141 142	Yes	Addendum Short term financial scheme	Together with UD.1 or UD.4	na
Supervisory measures	SM.4	4.4.4	Withdrawal or lapse of authorisation	Home NSA	Host NSA(s) EIOPA	144	Yes	Addendum Withdraw of authorisation	Together with UD.1 or UD.4	na

	Ref.	§ nr.	Information flow	Sender	Receiver	Solvency II article	Template if in scope	<u> </u>	Timing to reply
Supervisory measures	SM.5	4.4.5	Application of reorganisation measures	Home NSA	AII NSA	269 270 273 296		Together with UD.1 or UD.4	na

ANNEX 3: Information included in the templates used for notifications

The paragraphs below contain the list of information to be included in the template used for notifications in the IT tool. Those templates will be translated into *forms* for the exchange of information in the IT tool.

Intention to authorise (A.5)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking	2.6.3.1 A
2	The LEI of the undertaking (or if not available, the identification code used by the Supervisory Authority)	2.6.3.1 B
3	Where applicable, the name and addressinformation of the establishments (other than the head office of the insurance undertaking) and/or the branches ,or other establishments situated in the host Member States, other than the Home Member State, from which it plans to provide services	2.6.3.1 C
4	The classes of insurance according to the Annexes I and II to the Solvency II Directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive which the undertaking is applying for authorisation with indication for which those there is the intention to start cross-border business	2.6.3.1 D
5	A summary of the undertaking scheme of operations indicating: (a) the nature of the risks or commitments which the insurance undertaking proposes to cover in the Host Member State(s)	2.6.3.1 E.a
6	A summary of the undertaking scheme of operations indicating: (b) if available, the: projected gross written premiums, estimates of management expenses, general expenses and commissions; guiding principles of the reinsurance retrocession of the business, projected loss ratio of the portfolio and other projected technical indicators if available, for each Host Member State	2.6.3.1 E.b
7	A summary of the undertaking scheme of operations indicating: (c) if available, any information regarding expected local third or related parties involved in the underwriting activities in the Host Member State including but not limited to the foreseen distribution partners in the host Member State	2.6.3.1 E.c
8	Where the undertaking has clearly indicated its intention to operate exclusively, or almost exclusively, in the Host Member State(s): (a) a identification, through for example the person(s) ID card, passport or nomination document(s), of the persons who effectively run the undertaking or are responsible for the key functions;	2.6.3.1 F.a
9	Where the undertaking has clearly indicated its intention to operate exclusively, or almost exclusively, in the Host Member State(s): (b) a summary of the undertaking's system of governance, including the risk management system in place;	2.6.3.1 F.b
10	Where the undertaking has clearly indicated its intention to operate exclusively, or almost exclusively, in the Host Member State(s): (c) any available information resulting from the discussions with the undertaking of its business strategy and how the activities in the Host Member State fits into that strategy.	2.6.3.1 F.c
11	Field to attach documentation	NA
12	Free text field	NA
13	Possibility to combine this notification template with IB.2, RB.2 and/or F.2	2.6.3.2

Answer from Host NSA (A.6)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking	NA
2	The LEI of the undertaking (or if not available, the identification code used by the Supervisory Authority)	NA
3	Where applicable, the name and addressinformation of the establishments (other than the head office of the insurance undertaking) and/or the branches ,or other establishments situated in the host Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the Annexes I and II to the Solvency II Directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive which the undertaking is applying for authorisation with indication for which those there is the intention to start cross-border business	NA
5	In cases where an insurance undertaking applying for an authorisation has clearly indicated its intention to operate exclusively or almost exclusively in one or more input to facilitate the understanding of the situation and the circumstances of the undertaking by the Home NSA	2.6.4.2
6	Field to attach documentation	NA
7	Free text field	NA

EIOPA's view (A.7, IB.6, F.6, RB.3)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used by the Supervisory Authority)	NA
3	Where applicable, the name and addressinformation of the establishments (other than the head office of the insurance undertaking) and/or the branches ,or other establishments situated in the host Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the Annexes I and II to the Solvency II Directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive which the undertaking is applying for authorisation (is authorised) with indication for which those there is the intention to start (pursue) cross-border business	NA
5	Feedback on the contents of notification (taking into account the risk profile of the undertaking and/or the type of cross-border business the undertaking aims to underwrite)	2.6.4.3 3.1.2.5 3.2.2.4
6	Field to attach documentation	NA
7	Free text field	NA

End of authorisation process (A.8)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used by the Supervisory Authority)	NA
3	Where applicable, the name and addressinformation of the establishments (other than the head office of the insurance undertaking) and/or the branches ,or other establishments situated in the host Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the Annexes I and II to the Solvency II Directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive which the undertaking is applying for authorisation with indication for which those there is the intention to start cross-border business	NA

#	Information to be included in the template	§ nr.
5	Results of the decision making process to grant the authorisation to the undertaking	2.6.5.5
6	Field to attach documentation	NA
7	Free text field	NA

Communication of notification insurance Branch (IB.2)

#	Information to be included in the template	§ nr.
1	The name and address of the head office of the insurance undertaking	3.1.1.1 A
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	3.1.1.2 A
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	3.1.1.1 B
4	The classes of insurance according to the annexes I and II to the Solvency II Directive, for which the undertaking is authorised and into which the planned business falls	3.1.1.1 C
5	The scheme of operations setting out, at least, (a) the nature of the risks or commitments which the undertaking is proposing to cover through the branch	3.1.1.1 D.a
6	The scheme of operations setting out, at least, (b) if available, the guiding principles as to reinsurance and to retrocession with respect to the branch operations	3.1.1.1 D.b
7	The scheme of operations setting out, at least, (c) if available, cost estimates for setting up the administrative services and organisation for securing business of the branch; the financial resources set aside for this purpose and, if the risks to be covered fall within class n°18 in part a of annex I to the solvency II directive, the company taking over assistance services or the resources available to the insurance undertaking to provide the promised assistance	3.1.1.1 D.c
8	The scheme of operations setting out, at least, (d) the organisational structure of the branch	3.1.1.1 D.d
9	The scheme of operations setting out, at least, (e) if available, for the first three financial years of the branch: • estimates of management expenses, other than installation costs. in particular, general expenses and commissions; • estimates of premiums or contributions and claims;	3.1.1.1 D.e
10	The name of a person, and if available the e-mail address, who possesses sufficient powers to bind, in relation to third parties, the insurance undertaking or, in the case of Lloyd's, the underwriters concerned and to represent it or them in relations with the authorities and courts of the host member state (the authorised agent) and description of their powers;	3.1.1.1 E
10	If the insurance undertaking is to cover risks in class 10 in part a of annex i to the solvency ii directive, not including carrier's liability, a declaration that the undertaking has become a member of the national bureau and the national guarantee fund of the Host Member State;	3.1.1.1 F
11	If the insurance undertaking is to cover risks in class 10 in part a of annex i to the solvency ii directive, not including carrier's liability, the name and address of the claims representative as referred to in article 18(1)(h)	NA
12	In cases where the insurance undertaking intends to cover risks relating to legal expenses insurance, the option chosen from those described in article 200 of the solvency ii directive;	3.1.1.1 G
13	A certificate attesting that the insurance undertaking covers the SCR and MCR as calculated in accordance with articles 100 to 129 of the solvency ii directive in the form provided for in annex 1 hereto.	3.1.1.1 H
14	If available, identification of the person(s) who effectively run the branch or are responsible for key functions for the branch (persons responsible for the activities of the branch within the Home Member State), through for example the person(s) ID card, passport or nomination document(s) and if so required by the host Member State legislation, her address of residence in the host Member State	3.1.1.2 B
15	If the undertaking belongs to a cross-border group, the name of the group supervisor and the structure of the group as included in the coordination arrangement's annex, together with the last reported group solvency position;	3.1.1.2 C
16	Any available information regarding the planned distribution channel(s), relevant outsourcing contracts and partners that will be used in the Host Member State	3.1.1.2 D
17	If available the results of the assessment performed by the Home Authority of the undertaking's compliance with the pProduct oOversight and gGovernance requirements	3.1.1.2 E

#	Information to be included in the template	§ nr.
	set by the Commission Delegated Regulation (EU) 2017/2358 supplementing Directive (EU) 2016/97 of the European Parliament and of the Council (e.g.examples of the information which might be notified are the following: material findingds regarding design of the product, monitoring of distributors, target market identification and product testing)	
	A description of the relevant policyholder guarantee funds in the Home Member State, where applicable.	3.1.1.2 F
	A summary of the undertaking's system of governance, including the risk management system in place, which provides for the proper management of the business of the branch.	3.1.1.3 A
21	Any available information resulting from the discussions with the undertaking of its business strategy and how the branch fits into that strategy.	3.1.1.3 B
22	Field to attach documentation	NA
23	Free text field	NA

Ad hoc request from Host NSA to communication of notification insurance branch (IB.3)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the planned business falls	NA
5	Requests of ad hoc information from Host NSA. This field includes the areas of the Home NSA notification where the information is considered to be incomplete by Host NSA	3.1.1.4 3.1.2.2
6	Field to attach documentation	NA
7	Free text field	NA

Informal exchanges before the Home NSA sends the complete notification (IB.4)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the planned business falls	NA
5	Field to attach documentation	NA
6	Free text field	NA

Answer from Host NSA general answer to communication of notification insurance branch (IB.5)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Any conditions under which, in the interest of the general good, the activity must be pursued within the territory of the Host Member State or confirm that no conditions have been imposed in accordance with Article 146(3) of the Solvency II Directive.	3.1.2.1 A
	This information shall include the link to the website where the general good conditions are published	
6	Irregularities known to the Host NSA about the planned outsourcing activities, distribution partners, claims representatives, key persons as well as any relevant information following the analysis of the notification received from the Home NSA;	3.1.2.1 B
7	If the insurance undertaking or its parent undertaking tried to establish an insurance undertaking in the Host Member State. Where an application for authorisation had been declined, the Host NSA shall provide additional information	3.1.2.1 C
8	If a related insurance undertaking, within the meaning of Article 212 of the Solvency II Directive, of the undertaking or its parent undertaking previously established in the Host NSA had its authorisation revoked or withdrawn and the reasons supporting such a decision	3.1.2.1 D
9	Description of relevant policyholder guarantee funds in the Host Member State that would be applicable to the branch	3.1.2.1 E
10	Field to attach documentation	NA
11	Free text field	NA

Changes to insurance branch (IB.9)

#	Information to be included in the template	§ nr.
1	Changes to the name and address of the insurance undertaking;	NA
2	Changes to the LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the cross-border business falls	NA
4	Changes to the address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	3.1.5.1
5	Changes to the scheme of operations of the undertaking including the nature of the risks or commitments which the undertaking is proposing to cover through the branch	3.1.5.1
6	Changes to the scheme of operations of the undertaking including the guiding principles as to reinsurance and to retrocession with respect to the branch operations	3.1.5.1
7	Changes to the scheme of operations of the undertaking including the organisational structure of the branch	3.1.5.1
8	Changes to the name of a person, and if available the e-mail address, who possesses sufficient powers to bind, in relation to third parties, the insurance undertaking or, in the case of Lloyd's, the underwriters concerned and to represent it or them in relations with the authorities and courts of the host member state (the authorised agent) and description of their powers;	3.1.5.1
10	A certificate attesting that the insurance undertaking covers the SCR and MCR as calculated in accordance with articles 100 to 129 of the solvency ii directive in the form provided for in annex 1 hereto.	3.1.5.2

#	Information to be included in the template	§ nr.
11	Field to attach documentation	NA
12	Free text field	NA

Insurance branch discontinuation (IB.10)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Field to describe how the policy underwritten by the Branch will be managed (including the management of the run-off, if applicable)	3.1.6.1 3.1.6.2
6	Field to attach documentation	NA
7	Free text field	NA

Insurance branch closure (IB.11)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Confirmation of the closure of the insurance branch	3.1.6.4
	Field to describe how the policy underwritten by the Branch will be managed (including the management of the run-off, if applicable)	3.1.6.4
7	Field to attach documentation	NA
8	Free text field	NA

Confirmation of closure/discontinuation (IB.12, RB.7, F.13)

#	Information to be included in the template	§ nr.
1	The name and address of the insurance undertaking;	NA
2	The LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the branch in the host member state, from which documents, including all communications to the authorised agent, may be obtained or delivered to	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Confirmation of the receipt of the closure/discontinuation	NA
	Field to describe how the policy underwritten by the Branch will be managed (including the management of the run-off, if applicable)	NA
7	Field to attach documentation	NA
8	Free text field	NA

Communication of notification reinsurance branch (RB.2)

#	Information to be included in the template	§ nr.
1	The name and address of the head office of the undertaking	3.1.7.1 A
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the of the branch, which shall also be that of the authorised agent	3.1.7.1 B
4	The name and powers of the authorised agent	3.1.7.1 C
5	The type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised into which the planned business falls;	3.1.7.1 D
6	A certificate attesting that the reinsurance undertaking covers the SCR and MCR as calculated in accordance with Articles 100 to 129 of the Solvency II Directive, in the form provided for in Annex 1 hereto.	3.1.7.1 E
7	Field to attach documentation	NA
8	Free text field	NA

Answer from Host NSA to communication of notification reinsurance branch (RB.3)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the of the branch, which shall also be that of the authorised agent	NA
4	The name and powers of the authorised agent	NA
5	The type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised into which the planned business falls;	NA
6	Field to attach documentation	NA
7	Free text field	NA

Changes to reinsurance branch (RB.5)

#	Information to be included in the template	§ nr.
1	Changes to the name and address of the head office of the undertaking	3.1.7.1 A
2	Changes to the LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	Changes to the address of the of the branch, which shall also be that of the authorised agent	3.1.7.1 B
4	Changes to the name and powers of the authorised agent	3.1.7.1 C
5	The type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised into which the cross-border business falls	3.1.7.1 D
6	Certificate attesting that the reinsurance undertaking covers the SCR and MCR as calculated in accordance with Articles 100 to 129 of the Solvency II Directive, in the form provided for in Annex 1 hereto.	
7	Field to attach documentation	NA
8	Free text field	NA

Reinsurance branch discontinuation (RB.6)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the of the branch, which shall also be that of the authorised agent	NA
4	The type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised into which the cross-border business falls	NA
5	Information on the planned closure	NA
5	Field to attach documentation	NA
6	Free text field	NA

Reinsurance branch closure (RB.7)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The address of the of the branch, which shall also be that of the authorised agent	NA
4	The type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised into which the cross-border business falls	NA
5	Confirmation of the closure of the reinsurance branch	3.1.7.5
6	Field to describe how the policy underwritten by the Branch will be managed (including the management of the run-off, if applicable)	NA
7	Field to attach documentation	NA
8	Free text field	NA

Communication of notification to start FoS (F.2)

#	Information to be included in the template	§ nr.
1	The name and address of the head office of the undertaking	3.2.1.1 A
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	3.2.1.2 A
3	If the undertaking belongs to a cross-border group, the name of the group supervisor and the structure of the group as included in the coordination arrangement's annex, together with the last reported group solvency position	3.2.1.2 B
4	Any available information regarding expected local third or related parties involved in the underwriting activities, including but not limited to the foreseen distribution partners in the Host Member State	3.2.1.2 C
5	Identification, through for example the person's ID card, passport or nomination document(s) and an e-mail address, of the person who is responsible within the insurance undertaking for handling of complaints in relation to the FoS activities	3.2.1.2 D
6	If available the results of the assessment performed by the Home Authority of the undertaking's compliance with the Product Oversight and Governance requirements set by the Commission Delegated Regulation (EU) 2017/2358 supplementing Directive (EU) 2016/97 of the European Parliament and of the Council (examples of the information which might be notified are the following: material findings regarding design of the product, monitoring of distributors, target market identification and product testing	3.2.1.2 E
7	Description of the relevant policyholder guarantee funds in the Home Member State, where applicable	3.2.1.2 F
8	Where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services;	3.2.1.1 B

#	Information to be included in the template	§ nr.
9	The classes of insurance according to the annexes I and II to the Solvency II Directive, which the insurance undertaking has been authorised to offer	3.2.1.1 C
10	The nature of the risks or commitments which the insurance undertaking proposes to cover in the Host Member State	3.2.1.1 D
11	If the insurance undertaking intends to cover risks in class 10 in Part A of Annex I to the Solvency II Directive not including carrier's liability: (i) the name and address of the claims representative as referred to in Article 18(1)(h)	3.2.1.1 E.i
12	If the insurance undertaking intends to cover risks in class 10 in Part A of Annex I to the Solvency II Directive not including carrier's liability: (ii) a declaration that it has become a member of the national bureau and national guarantee fund of the Host Member State	3.2.1.1 E.ii
13	If the insurance undertaking intends to cover risks relating to legal expenses insurance, the option chosen from those described in Article 200 of the Solvency II Directive	3.2.1.1 F
14	A certificate attesting that the insurance undertaking covers the SCR and MCR calculated in accordance with Articles 100 and 129 of the Solvency II Directive, in the form provided in Annex I hereto	3.2.1.1 G
15	Where the undertaking intends to operate exclusively, or almost exclusively in other Member States, identification, through for example the persons(s) ID card, passport or nomination document(s), of the persons who effectively run the undertaking or are responsible for the key functions;	3.2.1.3 A
16	Where the undertaking intends to operate exclusively, or almost exclusively in other Member States, a summary of the undertaking's system of governance, including the risk management system in place	3.2.1.3 B
17	Where the undertaking intends to operate exclusively, or almost exclusively in other Member States, any available information resulting from the discussions with the undertaking of its business strategy and how the FoS fits into that strategy	3.2.1.3 C
18	Field to attach documentation	NA
19	Free text field	NA

Ad hoc request from Host NSA to communication of notification to start FoS (F.3)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	Where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, which the insurance undertaking has been authorised to offer	NA
5	Requests of ad hoc information from Host NSA. This field includes the areas of the Home NSA notification where the information is considered to be incomplete by Host NSA	3.2.1.4 3.2.2.2
6	Field to attach documentation	NA
7	Free text field	NA

Answer from Host NSA general answer to communication of notification to start FoS (F.5)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	Where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, which the insurance undertaking has been authorised to offer	NA
5	Any conditions under which, in the interest of the general good, the activity must be pursued within the territory of the Host Member State or confirm that no conditions have been imposed. This information shall include the link to the website where the general good conditions are published	3.2.2.1 A
6	Irregularities known to the Host NSA about the planned outsourcing activities, distribution partners, claims representatives, key persons as well as any relevant information following the analysis of the notification received from the Home NSA;	3.2.2.1 B
7	If the insurance undertaking or its parent undertaking tried to establish an insurance undertaking in the Host Member State. Where an application for authorisation had been declined, the Host NSA shall provide additional information	3.2.2.1 C
8	If a related insurance undertaking, within the meaning of Article 212 of the Solvency II Directive, of the undertaking or its parent undertaking previously established in the Host NSA had its authorisation revoked or withdrawn and the reasons supporting such a decision	3.2.2.1 D
9	Description of relevant policyholder guarantee funds in the Host Member State that would be applicable to the branch	3.2.2.1 E
10	Field to attach documentation	NA
11	Free text field	NA

Changes to freedom of services (F.10)

	• • • • • • • • • • • • • • • • • • • •	
#	Information to be included in the template	§ nr.
1	Changes to the name and address of the undertaking;	NA
2	Changes to LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	Changes to the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, which the insurance undertaking has been authorised to offer	NA
5	Changes to the nature of the risks or commitments which the insurance undertaking proposes to cover in the Host Member State	3.2.5.1
6	Field to attach documentation	NA
7	Free text field	NA

Freedom of services discontinuation start (F.11)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the Annexes I and II to the Solvency II Directive into which the planned business falls	NA
5	Information on the planned discontinuation	NA
6	Field to attach documentation	NA
7	Free text field	NA

Freedom of services discontinuation end (F.12)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	Where applicable, the name and address of the establishments (other than the head office of the insurance undertaking), situated in the Member States, other than the Home Member State, from which it plans to provide services	NA
4	The classes of insurance according to the annexes I and II to the solvency ii directive, which the insurance undertaking has been authorised to offer	NA
5	Confirmation of the effective discontinuation of the insurance activities	3.2.6.2
6	Field to describe how the policy underwritten by the undertaking in FoS will be managed (including the management of the run-off, if applicable)	3.2.6.2
7	Field to attach documentation	NA
8	Free text field	NA

Deteriorating of financial conditions or other emerging risks (UD.1)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	4.2.1.8 A
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	A description of the event(s) which have occurred to the insurance undertaking and a summary of the first Home NSA assessment, including where appropriate detailed documentation to understand the context	4.2.1.8 B
6	The supervisory measures ³⁴ requested by the Home NSA to the undertaking and the timeline for such measures to be completed in accordance to Section 4.5;	4.2.1.8 C
7	When available, an overview of the undertaking financial and solvency position, including where appropriate, a forward looking projection off the impacts of the supervisory measures requested by the Home NSA;	4.2.1.8 D
8	Where appropriate, the Home NSA reasoned request to the Host NSA(s) to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform on the undertaking;	4.2.1.8 E
9	The Home NSA request of information to the Host NSA(s) needed to perform its supervisory activities on the undertaking;	4.2.1.8 F
10	Other additional information and specifications as deemed necessary by the Home NSA	4.2.1.8 G
11	Field to attach documentation	NA
12	Free text field	NA

Reply to deteriorating financial conditions or other emerging risks (UD.2, UD.3)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	Where applicable, the LEI of the undertaking intending to pursue business under FoS (or if not available, the identification code used in the local market allocated by the Supervisory Authority);	NA
	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised and into which the cross-border business falls	NA
6	The request to the Home NSA to provide further information on the (notified) undertaking and or/the supervisory measures taken by the Home NSA	4.2.1.11
7	Field to attach documentation	NA
8	Free text field	NA

⁻

³⁴ The supervisory measures include the administrative sanctions or other measures imposed by the NSA against the undertaking highlighting those which can suspend the activity without suspending the pass-porting rights

Relevant updates on notifications of deteriorating financial conditions or other emerging risks (UD.4)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	4.2.1.8 A
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive,, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Update on the event(s) which have occurred to the insurance undertaking and a summary of the first Home NSA assessment, including where appropriate detailed documentation to understand the context	NA
6	Update on the supervisory measures requested by the Home NSA to the undertaking and the timeline for such measures to be completed in accordance to Section 4.5;	NA
7	When available, update on the overview of the undertaking financial and solvency position, including where appropriate, a forward looking projection off the impacts of the supervisory measures requested by the Home NSA;	NA
8	Where appropriate, the Home NSA reasoned request to the Host NSA(s) to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform on the undertaking;	NA
9	The Home NSA request of information to the Host NSA(s) needed to perform its supervisory activities on the undertaking;	NA
10	Other additional information and specifications as deemed necessary by the Home NSA	NA
11	The undertaking has resolved its difficulties	4.2.1.12
12	Field to attach documentation	NA
13	Free text field	NA

Host NSA's Concerns consumer protection (CCP.1)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	4.2.3.5 A
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	4.2.3.5 B
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	A description of the event(s) occurred to the insurance undertaking and a summary of the Host NSA assessment of the situation, including where appropriate any documentation that can clarify the issue	4.2.3.5 C
6	The supervisory measures requested by the Host NSA to the undertaking and their timeline of application;	4.2.3.5 D
7	Where appropriate, the Host NSA reasoned request to the Home NSA to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform on the undertaking	4.2.3.5 E
8	The Host NSA request of information to the Home NSA needed to perform its supervisory activities on the undertaking;	4.2.3.5 F
9	Other additional information and specifications as deemed necessary and proportionate by the Host NSA	4.2.3.5 G
10	Field to attach documentation	
11	Free text field	NA

Reply to concerns on consumer protection (CCP.2, CCP.3)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised and into which the cross-border business falls	NA
6	The request to the Host NSA to provide further information on the (notified) undertaking and or/the supervisory measures taken by the Home NSA	4.2.3.8
7	Field to attach documentation	NA
8	Free text field	NA

Relevant updates on concerns on consumer protection (CPP.4)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Update of the event(s) occurred to the insurance undertaking and a summary of the updated Host NSA assessment of the situation, including where appropriate any updated documentation that can clarify the issue	NA
6	Updates on the supervisory measures requested by the Host NSA to the undertaking and their timeline of application;	NA
7	Where appropriate, the Host NSA reasoned request to the Home NSA to set up collaboration platform and/or to EIOPA to consider setting up of collaboration platform on the undertaking	NA
8	The Host NSA request of information to the Home NSA needed to perform its supervisory activities on the undertaking;	NA
9	Other additional information and specifications as deemed necessary and proportionate by the Host NSA	NA
10	Field to attach documentation	4.2.3.9
11	Free text field	NA

Set up a platform of collaboration (CP.1)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II Directive,, for which the undertaking is authorised and into which the cross-border business falls	NA
5	Information on setting up a collaboration platform on the given undertaking	4.3.2.1
6	Field to attach documentation	NA
7	Free text field	NA

Update on platform of collaboration (CP.2)

#	Information to be included in the template	§ nr.
1	The name and address of the undertaking;	NA
2	LEI of the undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	The name and address of the establishments (other than the head office of the insurance undertaking) and/or the branches, situated in the Member States, other than the Home Member State, from which it provides services	NA
4	The classes of insurance according to the annexes I and II to the solvency II directive and/or the type of reinsurance activity, according to Article 15(5) of the Solvency II	NA

#	Information to be included in the template	§ nr.
	Directive, for which the undertaking is authorised and into which the cross-border business falls	
5	Relevant updates on the collaboration platform on the given undertaking	4.3.2.1
6	Field to attach documentation	NA
7	Free text field	NA

Portfolio transfer info to Host NSA of the Branch whose portfolio is to be transferred (PT.2)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	4.4.1.4 B.a
2	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The scope of the operation (total or partial transfer of the branch's portfolio)	4.4.1.4 B.a
4	The classes of insurance according into which the portfolio to be transferred falls	4.4.1.4 B.a
5	the details of the nature of the risks or commitments to be transferred	4.4.1.4 B.a
6	The name and address of the accepting undertaking	4.4.1.4 B.a
7	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
8	The draft transfer agreement or the transfer agreement.	4.4.1.4 B.a
9	Arrangements for the settlement of claims in the event of the closure of the branch following the transfer (LINK TO IB.11 and RB.7)	4.4.1.4.B.b
10	Request for information (consultation) to the Host NSA	4.4.1.1 A
	Request to agree to the portfolio transfer, in case of risks and commitments in that member state	4.4.1.1 B
11	Field to attach documentation	NA
12	Free text field	NA

Portfolio transfer info to NSA of the Member State of the risks or commitments (PT.3)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	4.4.1.4 C.a
2	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	The name and address of the accepting undertaking	4.4.1.4 C.a
	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	The draft transfer agreement or the transfer agreement.	4.4.1.4 C.a
	Request to agree to the portfolio transfer	4.4.1.1 B
	Field to attach documentation	NA
	Free text field	NA

Portfolio transfer Host NSAs opinion/consent reply (PT.4)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	NA
	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting undertaking	NA
	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA

#	Information to be included in the template	§ nr.
5	Opinion/consent to the portfolio transfer	4.4.1.2
	Circumstances and the period within which contracts within the transferred portfolio may be cancelled according to the provisions of their national law	4.4.1.10
7	Field to attach documentation	NA
8	Free text field	NA

Portfolio transfer info to Home NSA of the accepting (re)insurance undertaking (PT.5)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	4.4.1.4 A.a
2	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The classes of insurance according into which the portfolio to be transferred falls	4.4.1.4 A.a
4	the details of the nature of the risks or commitments to be transferred	4.4.1.4 A.a
5	The name and address of the accepting undertaking	4.4.1.4 A.a
6	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
7	The draft transfer agreement or the transfer agreement.	4.4.1.4 A.a
8	The volume of gross and net technical provisions, established on the basis of the contracts to be transferred	4.4.1.4 A.b
9	The volume of gross and net written premiums	4.4.1.4 A.c
10	The volume of the gross and net burden of claims in non-life insurance	4.4.1.4 A.d
11	Details of assets transferred	4.4.1.4 A.e
12	Details of guarantees provided by the transferring insurance undertaking or a third party (for example a reinsurance undertaking) to safeguard against deterioration of the reserves corresponding to the transferred business	4.4.1.4 A.f
13	Name(s) of the country or countries of the risks or commitments	4.4.1.4 A.g
14	Request of a certificate attesting that the accepting undertaking covers the SCR and MCR calculated in accordance with Articles 100 and 129 of the Solvency II Directive taking into account the transfer, in the form provided in Annex I hereto	4.4.1.3
15	Field to attach documentation	NA
16	Free text field	NA

Portfolio transfer Home NSA of the accepting (re)insurance undertaking reply (PT.6)

_	-	
#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	NA
	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting undertaking	NA
4	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
5	Concerns about how the accepting undertaking will perform in the future	4.4.1.5
6	Certificate attesting that the accepting undertaking covers the SCR and MCR calculated in accordance with Articles 100 and 129 of the Solvency II Directive taking into account the transfer, in the form provided in Annex I hereto	
7	Circumstances and the period within which contracts within the transferred portfolio may be cancelled according to the provisions of their national law	4.4.1.10
8	Field to attach documentation	NA

#	 Information to be included in the template	§ nr.
9	Free text field	NA

Communication of the decision of portfolio transfer (PT.7, PTTC.5)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	NA
	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting undertaking	NA
4	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
5	Communication of the decision regarding the portfolio transfer to the Supervisory Authority of the country or countries where the contracts were concluded and the other authorities which were consulted	4.4.1.7
6	Field to attach documentation	NA
7	Free text field	NA

Communication of the method of publication of portfolio transfer (PT.8, PTTC.6)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring undertaking	NA
	LEI of the transferring undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting undertaking	NA
4	LEI of the accepting undertaking (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
	Communication of the method of publication of the transfer of portfolio provided by their national law	4.4.1.7
6	Field to attach documentation	NA
7	Free text field	NA

Requests to the NSA of the accepting office (PTTC.1)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring third country branch established in the Transferor territory	NA
2	LEI of the transferring third country branch (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting office	NA
4	LEI of the accepting officer* (or if not available, the identification code used in the local market allocated by the Supervisory Authority) * accepting officer can be an undertaking, a branch of an undertaking or a third country branch	NA
	Request to the NSA of the accepting office to certify that accepting office possesses the necessary solvency	4.4.3.1 A-C
	Request to the NSA of the accepting office to certify that the law of the Member State of the accepting office permits such a transfer and that the Member State has agreed to the transfer	4.4.3.1 D
7	Field to attach documentation	NA
8	Free text field	NA

Requests to the NSA of the Member State of the risks or commitments (PTTC.2)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring third country branch established in the Transferor territory	NA
2	LEI of the transferring third country branch (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting office	NA
4	LEI of the accepting officer* (or if not available, the identification code used in the local market allocated by the Supervisory Authority) * accepting officer can be an undertaking, a branch of an undertaking or a third country branch	
5	Request of an opinion in relation to the portfolio transfer	4.4.3.1 E
6	Field to attach documentation	NA
7	Free text field	NA

Reply of the NSA of the accepting office (PTTC.3)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring third country branch established in the Transferor territory	NA
2	LEI of the transferring third country branch (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting office	NA
4	LEI of the accepting officer* (or if not available, the identification code used in the local market allocated by the Supervisory Authority) * accepting officer can be an undertaking, a branch of an undertaking or a third country branch	
5	Certification that accepting office possesses the necessary solvency	4.4.3.1 A-C
6	Certification that the law of the Member State of the accepting office permits such a transfer and that the Member State has agreed to the transfer	4.4.3.1 D
7	Field to attach documentation	NA
8	Free text field	NA

Reply of the NSA of the Member State of the risks or commitments (PTTC.4)

#	Information to be included in the template	§ nr.
1	The name and address of the transferring third country branch established in the Transferor territory	NA
2	LEI of the transferring third country branch (or if not available, the identification code used in the local market allocated by the Supervisory Authority)	NA
3	The name and address of the accepting office	NA
4	LEI of the accepting officer* (or if not available, the identification code used in the local market allocated by the Supervisory Authority) * accepting officer can be an undertaking, a branch of an undertaking or a third country branch	
5	Opinion in relation to the portfolio transfer	4.4.3.1 E
6	Field to attach documentation	NA
7	Free text field	NA

EIOPA

Westhafen Tower, Westhafenplatz 1
60327 Frankfurt – Germany
Tel. + 49 69-951119-20
info@eiopa.europa.eu
https://www.eiopa.europa.eu